DISCUSSION PAPER SERIES

No. 8610

FISCAL MULTIPLIERS OVER THE BUSINESS CYCLE

Pascal Michaillat

INTERNATIONAL MACROECONOMICS



Centre for Economic Policy Research

www.cepr.org

www.cepr.org/pubs/dps/DP8610.asp

Available online at:

FISCAL MULTIPLIERS OVER THE BUSINESS CYCLE

Pascal Michaillat, London School of Economics and CEPR

Discussion Paper No. 8610 October 2011

Centre for Economic Policy Research 77 Bastwick Street, London EC1V 3PZ, UK Tel: (44 20) 7183 8801, Fax: (44 20) 7183 8820 Email: cepr@cepr.org, Website: www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programme in **INTERNATIONAL MACROECONOMICS**. Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Pascal Michaillat

CEPR Discussion Paper No. 8610

October 2011

ABSTRACT

Fiscal Multipliers Over the Business Cycle*

This paper develops a theory characterizing the effects of fiscal policy on unemployment over the business cycle. The theory is based on a model of equilibrium unemployment in which jobs are rationed in recessions. Fiscal policy in the form of government spending on public-sector jobs reduces unemployment, especially during recessions: the fiscal multiplier---the reduction in unemployment rate achieved by spending one dollar on publicsector jobs---is positive and countercyclical. Although the labor market always sees vast flows of workers and a great deal of matching, recessions are periods of acute job shortage without much competition for workers among recruiting firms. Hence hiring in the public sector reduces unemployment effectively because it does not crowd out hiring in the private sector much. An implication is that empirical studies should control for the state of the economy when fiscal policies are implemented to estimate accurately the amplitude of fiscal multipliers in recessions.

JEL Classification: E24, E32, E62 and J64 Keywords: business cycle, fiscal multiplier, job rationing, matching frictions and unemployment

Pascal Michaillat Department of Economics, London School of Economics, Houghton Street, London WC2A 2AE

Email: p.michaillat@lse.ac.uk

For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=172893 * I am especially indebted to George Akerlof and Yuriy Gorodnichenko for their advice, numerous comments, and support. I thank Francesco Caselli, Pierre-Olivier Gourinchas, Ethan Ilzetzki, David Levine, Evi Pappa, Demian Pouzo, David Romer, and Emmanuel Saez, who provided valuable comments and suggestions. I am also grateful to numerous seminar participants for helpful discussions and comments. Financial support from the Center for Equitable Growth at UC Berkeley is gratefully acknowledged.

Submitted 05 October 2011

1 Introduction

This paper develops a theory characterizing the effects of fiscal policy on unemployment over the business cycle.¹ Such a theory is lacking from the macroeconomics literature because: (i) existing dynamic stochastic general equilibrium (DSGE) models used to study fiscal policy do not capture the job shortage that explains most of recessionary unemployment; and (ii) fiscal policy in these models does not stimulate aggregate demand for labor, and is bound to be ineffective in recessions.

In standard real business cycle (RBC) and new Keynesian (NK) models, the labor market always clears and there is no unemployment [Aiyagari et al., 1992; Baxter and King, 1993; Christiano and Eichenbaum, 1992; Christiano et al., 2011; Galí et al., 2007; Woodford, 2011]. These models are of little help to determine the effectiveness of fiscal policy at reducing unemployment.

Some RBC and NK models incorporate the model of unemployment of Pissarides [2000], in which frictions impede matching of recruiting firms with unemployed workers and wages are set by Nash bargaining between firms and workers [Andolfatto, 1996; Bruckner and Pappa, forthcoming; Merz, 1995; Monacelli et al., 2010]. These models have some unemployment but no recessions, because there are no labor market fluctuations. The absence of recessions results from the flexibility of bargained wages, which adjust fully to productivity shocks [Shimer, 2005]. Hence, these models cannot characterize the effects of fiscal policy in recessions.

¹This paper focuses on unemployment because, even though employment is not an exact measure of social welfare, the social cost of recessions ultimately comes from the increase in unemployment they generate: Sullivan and von Wachter [2009] document an increase in mortality caused by job displacements; von Wachter et al. [2009] document long-term earning losses following mass layoffs, which could reflect long-term losses in human capital; Hawton and Platt [2000] survey the medical literature studying suicidal behavior among unemployed workers; and Brenner and Mooney [1983] survey the medical literature measuring the health costs of unemployment. Accordingly, the first goal of policymakers in recessions is to reduce unemployment [for example, Romer and Bernstein, 2009].

Following Hall [2005], some NK models combine the Pissarides [2000] model of unemployment with rigid wages to generate labor market fluctuations and recessions [Blanchard and Galí, 2010; Gertler et al., 2008]. But all unemployment is frictional in these models as their labor market converges to full employment when matching frictions become arbitrarily small, or equivalently, when unemployed workers exert an arbitrarily large effort to search for jobs [Landais et al., 2010; Michaillat, forthcoming]. This property is at odd with the long queues of jobseekers in front of factory gates during the Great Depression. Thus these models are inadequate to study recessionary unemployment because they do not capture the job shortage observed in recessions.

Fiscal policy in these models typically consists of government purchases of consumption goods [Ramey, 2011].² Government consumption reduces the amount of goods available for private consumption, stimulating labor supply through a negative wealth effect (in the basic model with lump-sum taxes). But when jobs are rationed in recessions, unemployment remains high irrespective of labor supply [Landais et al., 2010]. Hence this policy is bound to be ineffective in recessions, and the analysis of government consumption is not informative of the scope for unemployment-reducing fiscal policy.

To study how fiscal policy can reduce recessionary unemployment, this paper proposes a barebone model with: (1) equilibrium unemployment as in Pissarides [2000]; (2) recessions driven by real wage rigidity as in Hall [2005]; (3) job rationing in recessions resulting from the combination of wage rigidity and diminishing marginal returns to labor as in Michaillat [forthcoming]; and (4) fiscal policy in the form of public-sector employment. The choice of public employment is

²There are a few exceptions: Finn [1998] or Cavallo [2005] consider government purchases of hours in the public sector; Baxter and King [1993] or Leeper et al. [2010] consider government investments.

guided by theory, as public employment stimulates aggregate labor demand, which is depressed in recessions; by empirical results, as government spending on public employment seems to have a higher output multiplier than government spending on investment or consumption goods [Pappa, 2010]; and by the relevance of the analysis for governments, who often use public employment to reduce unemployment in recessions: the Roosevelt administration hired millions of unemployed workers to build dams, bridges, and roads during the Great Depression [Fishback et al., 2003; Fleck, 1999]; the American Jobs Act presented by the Obama administration to Congress in 2011 proposes to spend \$140 billion to hire teachers and other public-sector workers.

The central result of the paper is that in a model with job rationing, the *public-employment multiplier*—the reduction in unemployment rate achieved by spending one dollar on public-sector jobs—is positive and countercyclical. Hence fiscal policy in the form of hiring in the public sector reduces unemployment, especially in recessions. This result hinges on the nature of unemployment in recessions: although the labor market always sees vast flows of workers and a great deal of matching activity, recessions are periods of acute job shortage without much competition for workers among recruiting firms. Hence hiring in the public sector reduces unemployment effectively because it does not crowd out hiring in the private sector much.

To understand the scope for fiscal policy, I underline necessary conditions for fiscal policy to reduce unemployment effectively. First, public employment is not effective in search-and-matching models without job rationing. Second, fiscal policies stimulating labor supply, such as monitoring unemployed workers' job search, have virtually no effect on unemployment when jobs are rationed in recessions. I also show that the *wage-subsidy multiplier*—the reduction in unemployment rate achieved by spending one dollar on a wage subsidy—is positive and countercyclical. A wage subsidy is effective here because technology shocks drive fluctuations and recessions arise from rigid wages being too high relative to labor productivity. But a wage subsidy may not be effective if demand shocks drive fluctuations, firms face a downward-sloping goods demand, and prices are completely sticky in the short run. In that case firm's labor demand is fully determined by goods demand and production function so that a wage subsidy is ineffective. On the contrary, public employment would reduce unemployment in any model with job rationing, independent of the source of fluctuations.

A critical implication of the paper is that available estimates of the fiscal multiplier do not apply in recessions, because the fiscal multiplier fluctuates widely over the business cycle while the estimation methods used in the literature average the effect of fiscal policy over all possible states of the economy [Parker, 2011].

The paper proceeds as follows. Section 2 presents a generic model of equilibrium unemployment with a public sector. Section 3 proves that the welfare-maximizing centralized allocation in this model is constant, irrespective of the stochastic process driving economic fluctuations. Hence unemployment fluctuations over the business cycle in the decentralized economy are always inefficient. Section 4 introduces recessions and job rationing into the model to prove that the publicemployment multiplier is positive and countercyclical. Section 5 examines the scope for (i) public employment in alternative models, and (ii) alternative fiscal policies in a model with job rationing. Section 6 concludes by discussing implications and limitations of the paper. Derivations and proofs are collected in the Appendix.

2 Model of Equilibrium Unemployment with a Public Sector

This section presents a generic model of equilibrium unemployment. Risk-averse unemployed workers search for a vacant job indiscriminately in the private and public sectors. Some frictions impede matching of jobseekers with vacancies. Labor market fluctuations are driven by technology, which follows a stochastic process $\{a_t\}_{t=0}^{+\infty}$.³ The time-*t* elements of the worker's choice, the firm's choice, and government policies are function of the history of technology levels $a^t \equiv (a_0, a_1, \dots, a_t)$ and of initial employments (g_{-1}, l_{-1}) in the public and private sectors.

2.1 Labor market

There is a unit mass of workers in the labor market. The labor market is composed of two sectors: a private sector with l_t workers, and a public sector with g_t workers. Public employment follows a stochastic process $\{g_t\}_{t=0}^{+\infty}$, whose time-*t* element is measurable with respect to (a^t, g_{-1}, l_{-1}) . Aggregate employment is $n_t = l_t + g_t$. Jobseekers apply to jobs randomly, not directing their search toward any of the two sectors.⁴

At the end of period t - 1, a fraction s of the n_{t-1} existing worker-job matches is exogenously destroyed. Workers who lose their job apply for a new job immediately. At the beginning of period t, $u_t = 1 - (1 - s) \cdot n_{t-1}$ unemployed workers search for a job with effort $e_t \in [0, 1]$. Those who find a job participate in production in period t with the $(1 - s) \cdot n_{t-1}$ incumbent workers.

³Empirical evidence suggests that recessions are driven by aggregate-activity shocks and not by reallocation shocks [Abraham and Katz, 1986; Blanchard and Diamond, 1989]. Hence I assume a stable matching function and introduce, in line with the literature, aggregate technology shocks.

⁴Models of the labor market do not usually include a public sector. Holmlund and Linden [1993], Quadrini and Trigari [2007], Hörner et al. [2007], and Gomes [2010] are exceptions that add a public sector to a search-and-matching model to study the macroeconomic impact of public-employment policies.

By posting a total of v_t vacancies, a representative firm hires workers in the private sector and the government hires workers in the public sector. The number h_t of matches made in the period is given by a matching function $h_t = h(e_t \cdot u_t, v_t)$ of aggregate search effort $e_t \cdot u_t$ and vacancies v_t , with the restriction that $h(e_t \cdot u_t, v_t) \le \min \{u_t, v_t\}$. To simplify the analysis, I specify a Cobb-Douglas matching function: $h(e_t \cdot u_t, v_t) = \omega_h \cdot (e_t \cdot u_t)^{\eta} \cdot v_t^{1-\eta}$, where ω_h and $\eta \in (0, 1)$ are parameters.

Labor market conditions are summarized by labor market tightness $\theta_t \equiv v_t/(e_t \cdot u_t)$. The matching technology is such that not all unemployed workers can find a job and not all vacancies can be filled. A jobseeker finds a job with probability $f(\theta_t) = h(e_t \cdot u_t, v_t)/(e_t \cdot u_t) = h(1, \theta_t)$ per unit of search effort; hence a jobseeker searching with effort e_t finds a job with probability $e_t \cdot f(\theta_t)$. Vacancies in the public and private sectors are filled with the same probability $q(\theta_t) = h(e_t \cdot u_t, v_t)/v_t = h(1/\theta_t, 1)$. In a tight market it is easy to find jobs—the per-unit jobfinding probability $f(\theta_t)$ is high—but difficult to find workers—the job-filling probability $q(\theta_t)$ is low.

Keeping a vacancy open has a per-period cost $r \cdot a_t$ in units of private good, where r captures resources spent recruiting workers.⁵ I assume away randomness at the firm level: a worker is hired with certainty by opening $1/q(\theta_t)$ vacancies and spending $r \cdot a_t/q(\theta_t)$.

⁵As in Pissarides [2000], the cost of opening a vacancy is proportional to technology a_t . This assumption can be justified on the grounds that recruiting is a labor-intensive activity so the cost to maintain a vacancy open depends on wages, which are nearly proportional to technology in practice. This assumption also simplifies derivations.

2.2 Wage

The wage is set once worker and firm have matched. Since search costs are sunk at the time of matching, there are always mutual gains from trade. There is no compelling theory of wage determination in such an environment [Hall, 2005]. Hence I assume that the equilibrium wage in the private sector follows a stochastic process $\{w_t\}_{t=0}^{+\infty}$, whose time-*t* element is measurable with respect to (a^t, g_{-1}, l_{-1}) . I impose that the wage process $\{w_t\}_{t=0}^{+\infty}$ be privately efficient: worker-firm pairs exploit all opportunities for mutual improvement. Private efficiency guarantees that the wage never causes the destruction of a match generating a positive bilateral surplus, a reasonable equilibrium requirement when rational workers and firms engage in long-term interactions [Barro, 1977]. Below, I consider as special cases of this wage process a rigid wage and the outcome of Nash bargaining. Finally, I assume that the government sets the public-sector wage at the level of the private-sector wage.

2.3 Worker

Workers have utility that depends on consumption c_t of private good produced in the private sector, consumption p_t of public good produced in the public sector, and job-search effort e_t , of the form $\chi \cdot \ln(p_t) + \ln(c_t) - k(e_t)$, where $k(e_t) = \omega_k \cdot e_t^{1+\kappa}/(1+\kappa)$. χ , ω_k , and $\kappa > 0$ are parameters. Job-search effort is not observable. Employed workers earn a wage w_t , and unemployed workers receive unemployment insurance (UI) benefits $b_t \cdot w_t$. Workers neither borrow nor save, so consumption is w_t when employed and $b_t \cdot w_t$ when unemployed. The benefit rate follows a stochastic process $\{b_t\}_{t=0}^{+\infty}$, whose time-*t* element is measurable with respect to (a^t, g_{-1}, l_{-1}) .

Given a benefit rate, wage, and labor market tightness $\{b_t, w_t, \theta_t\}_{t=0}^{+\infty}$, the *representative worker's problem* is to choose job-search effort $\{e_t\}_{t=0}^{+\infty}$ to maximize the expected utility

$$\mathbb{E}_0 \sum_{t=0}^{+\infty} \delta^t \cdot \left\{ (1-n_t^s) \cdot \ln(b_t \cdot w_t) + n_t^s \cdot \ln(w_t) - \left[1 - (1-s) \cdot n_{t-1}^s \right] \cdot k(e_t) \right\},\$$

subject to the law of motion of the probability n_t^s to be employed in period t

$$n_t^s = (1-s) \cdot n_{t-1}^s + [1 - (1-s) \cdot n_{t-1}^s] \cdot e_t \cdot f(\theta_t).$$

 \mathbb{E}_0 is the mathematical expectation conditional on time-0 information, δ is the discount factor.⁶ The first-order condition with respect to search effort e_t simplifies to

$$\left\{\frac{k'(e_t)}{f(\theta_t)} - \delta \cdot (1-s) \cdot \mathbb{E}_t\left[\frac{k'(e_{t+1})}{f(\theta_{t+1})}\right]\right\} + \kappa \cdot \delta \cdot (1-s) \cdot \mathbb{E}_t\left[k(e_{t+1})\right] = \ln\left(\frac{1}{b_t}\right).$$
(1)

2.4 Firm

The private sector is composed of a representative firm producing a private good sold to workers taking price as given. The firm's production function satisfies

$$y_t = a_t \cdot y(l_t) = a_t \cdot l_t^{\alpha}, \tag{2}$$

⁶The law of motion is an application of the total probability theorem. The probability n_t^s to have a job at *t* is the sum of the probability $e_t \cdot f(\theta_t)$ to find a job in *t* conditional on being jobless in t - 1, which occurs with probability $1 - n_{t-1}^s$, plus the probability $e_t \cdot f(\theta_t)$ to find a job in *t* conditional on losing a job at the end of t - 1, which occurs with probability $s \cdot n_{t-1}$, plus the probability of keeping a job at the end of t - 1, which occurs with probability $(1 - s) \cdot n_{t-1}^s$.

where y_t is output of private good, a_t is technology, l_t is employment in the firm, and α is a parameter. The firm is owned by risk-neutral entrepreneurs with the same discount factor δ as workers. Given technology, wage, and labor market tightness $\{a_t, w_t, \theta_t\}_{t=0}^{+\infty}$, the *representative firm's problem* is to choose employment $\{l_t\}_{t=0}^{+\infty}$ to maximize expected profit

$$\mathbb{E}_0 \sum_{t=0}^{+\infty} \delta^t \cdot \left\{ a_t \cdot y(l_t) - w_t \cdot l_t - \frac{r \cdot a_t}{q(\theta_t)} \cdot \left[l_t - (1-s) \cdot l_{t-1} \right] \right\}.$$

Since endogenous layoffs and quits never occur, $l_t - (1 - s) \cdot l_{t-1} \ge 0$ is the number of hires in period *t*.⁷ The first-order condition with respect to employment l_t is

$$a_t \cdot y'(l_t) = w_t + \frac{r \cdot a_t}{q(\theta_t)} - \delta \cdot (1 - s) \cdot \mathbb{E}_t \left[\frac{r \cdot a_{t+1}}{q(\theta_{t+1})} \right],$$
(3)

which implies that the firm hires labor until until marginal product of labor $a_t \cdot y'(l_t)$ equals marginal cost of labor, which is the sum of wage w_t , plus hiring cost $r \cdot a_t/q(\theta_t)$, minus discounted cost of hiring next period $\delta \cdot (1-s) \cdot \mathbb{E}_t [r \cdot a_{t+1}/q(\theta_{t+1})]$.

2.5 Equilibrium

Wages follow an exogenous stochastic process and cannot equalize labor supply and labor demand. Instead, labor market tightness θ_t equilibrates labor demand $(l_t + g_t)$ to labor supply n_t^s each period:

$$l_t + g_t = n_t^s = n_t. aga{4}$$

⁷There are no endogenous separations because the wage process is privately efficient. As in Michaillat [forthcoming], I can derive a sufficient condition for the wage process to be privately efficient.

Given a technology process $\{a_t\}_{t=0}^{+\infty}$, a wage process $\{w_t\}_{t=0}^{+\infty}$, a government policy $\{b_t, g_t\}_{t=0}^{+\infty}$, an *equilibrium with public sector* is a collection of stochastic processes $\{e_t, l_t, \theta_t\}_{t=0}^{+\infty}$ that solve the worker's problem (1), the firm's problem (3), and satisfy the equilibrium condition (4).

3 Efficient Allocation

This section solves the problem of a benevolent social planner who faces the technological constraints and labor market frictions present in the decentralized economy. The resulting allocation underscores that the welfare cost of recessions arises from the reduction in output and the increase in inequality due to unemployment.

3.1 Definitions

An *allocation* is a collection of stochastic processes $\{g_t, l_t, \theta_t, e_t, c_t^l, c_t^g, c_t^u, y_t, p_t\}_{t=0}^{+\infty}$ for public and private employment; labor market tightness; job-search effort; consumption in private jobs, public jobs, and unemployment; and output of private and public good; whose time-*t* element are measurable with respect to (a^t, l_{-1}, g_{-1}) . A *feasible allocation* is an allocation that satisfies (i) the production constraint (2) for the private good; (ii) a production constraint for the public good:

$$p_t = \mathbf{\omega}_p \cdot a_t \cdot \mathbf{y}(g_t), \tag{5}$$

where ω_p is a parameter that scales productivity in the public sector relative to that in the private sector; (iii) the resource constraint in the economy, which imposes that the private good be either

consumed or allocated to recruiting:

$$y_{t} = \left[l_{t} \cdot c_{t}^{l} + g_{t} \cdot c_{t}^{g} + (1 - n_{t}) \cdot c_{t}^{u}\right] + \frac{r \cdot a_{t}}{q(\theta_{t})} \cdot [n_{t} - (1 - s) \cdot n_{t-1}];$$
(6)

and (iv) the law of motion for aggregate employment

$$(1-s) \cdot n_{t-1} + u_t \cdot e_t \cdot f(\theta_t) = n_t, \tag{7}$$

where $n_t = l_t + g_t$. The *efficient allocation* is the feasible allocation that maximizes expected utility

$$\mathbb{E}_0 \sum_{t=0}^{+\infty} \delta^t \cdot \left\{ \chi \cdot \ln(pt) + \left[l_t \cdot \ln(c_t^l) + g_t \cdot \ln(c_t^g) + (1-n_t) \cdot \ln(c_t^u) \right] - \left[1 - (1-s) \cdot n_{t-1} \right] \cdot k(e_t) \right\}.$$

3.2 Properties

The first critical property of the efficient allocation is that there is perfect insurance against unemployment risk. Since workers are risk averse, it is optimal that all workers consume the same amount c_t of private good each period: $c_t = c_t^l = c_t^g = c_t^u$.

The second critical property of the efficient allocation, formalized in Proposition 1, is that labor market variables (public and private employment, job-search effort, and labor market tightness) are constant over time, independent of the technology process. Therefore there are no cyclical fluctuations on the labor market in the efficient allocation. In other words, unemployment fluctuations over the business cycle are socially inefficient.

PROPOSITION 1. In the efficient allocation, labor market variables $\{g_t, l_t, \theta_t, e_t\}_{t=0}^{+\infty}$ are deter-

ministic, independent of the technology process $\{a_t\}_{t=0}^{+\infty}$. These variables remain constant over time for an appropriate choice of initial employments l_{-1} and g_{-1} . The ratios of private consumption to technology $\{c_t/a_t\}_{t=0}^{+\infty}$, private output to technology $\{y_t/a_t\}_{t=0}^{+\infty}$, and public output to technology $\{p_t/a_t\}_{t=0}^{+\infty}$ share the same property.

I give an overview of the proof to illuminate the key economic mechanisms behind this property. First, the resource and production constraints imply that the ratios of private consumption, private output, and public output to technology are only function of current labor market variables:

$$c_t/a_t = y(l_t) - s \cdot \frac{r}{q(\theta_t)} \cdot n_t$$
$$y_t/a_t = y(l_t)$$
$$p_t/a_t = \omega_p \cdot y(g_t).$$

Therefore, these ratios remain constant over time if labor market variables do. Labor market variables in the efficient allocation are four stochastic processes $\{g_t, l_t, \theta_t, e_t\}_{t=0}^{+\infty}$ characterized by four equations, which I present below. Since hiring a worker has the same marginal cost in the private and public sector, the marginal benefit from a worker must be equal in both sectors:

$$\boldsymbol{\omega}_p \cdot \boldsymbol{\chi} \cdot \frac{c_t/a_t}{p_t/a_t} = \frac{y'(l_t)}{y'(g_t)} = \left[\frac{g_t}{l_t}\right]^{1-\alpha}$$

In particular, it is always optimal to employ some workers in the public sector as long as the public good is valuable ($\chi > 0$) and the government is productive ($\omega_p > 0$). The ratios p_t/a_t and c_t/a_t are only function of current labor market variables, so the efficient ratio of public to

private employment g_t/l_t is only function of current labor market variables. Intuitively, the tradeoff between private and public employment remains unchanged when technology falls because the productivities of public and private jobs fall in concert. Next, employments g_t and l_t , search effort e_t , and labor market tightness θ_t , are related by the law of motion of aggregate employment

$$(1-s) \cdot (l_{t-1}+g_{t-1}) + [1-(1-s) \cdot (l_{t-1}+g_{t-1})] \cdot e_t \cdot f(\theta_t) = (l_t+g_t),$$

which does not involve technology. The marginal cost of effort must equal its marginal benefit:

$$k'(e_t) = \frac{\eta}{1-\eta} \cdot r \cdot \theta_t \cdot \frac{1}{c_t/a_t}.$$

The efficient search effort e_t is only function of current labor market variables. Intuitively when technology falls, the marginal benefit of search remains unaffected because the vacancy-posting cost, expressed in utility terms, is unchanged (the vacancy-posting cost expressed in private-good terms falls, but the marginal utility of private-good consumption increases by the same amount). Finally, efficient labor market tightness θ_t satisfies

$$(1-\eta)\cdot y'(l_t) = \frac{r}{q(\theta_t)} - \delta \cdot (1-s) \cdot \mathbb{E}_t \left[\frac{c_t/a_t}{c_{t+1}/a_{t+1}} \cdot \left\{ \frac{r}{q(\theta_{t+1})} - \eta \cdot r \cdot \frac{\kappa}{1+\kappa} \cdot e_{t+1} \cdot \theta_{t+1} \right\} \right].$$

This relationship is similar to that in the canonical model of equilibrium unemployment, but for the factor $\kappa/(\kappa + 1) \cdot e_{t+1}$ [Pissarides, 2000, Chapter 1]. It says that the marginal benefit from having workers search for jobs must equal the marginal benefit from having them produce goods. It depends only on labor market variables and not on technology. Intuitively, when technology falls public and private sectors become less productive, but recruiting also becomes less costly so that the trade-off between production and search remains unchanged. This last equation completes the characterization of the efficient allocation. This system of four equations does not involve the technology process $\{a_t\}_{t=0}^{+\infty}$. Thus, the efficient allocation is deterministic, and it remains constant over time for an appropriate choice of initial employments l_{-1} and g_{-1} .

3.3 Implementation

Since job-search effort is not observable, the government cannot provide unemployment insurance contingent on search effort. As effort enters negatively in the utility function, unemployed workers would have no incentive to search if they were fully insured against unemployment risk. Hence full insurance would imply no effort and no employment, which is suboptimal. Because of moral hazard, the government must provide incentives to search by reducing the consumption of unemployed workers. Thus the efficient allocation, which provides full insurance against unemployment risk, cannot be implemented in a decentralized economy. In fact two inefficiencies arise in the decentralized economy: the government does not provide perfect unemployment insurance and unemployment may be above its efficient level. This paper focuses on the reduction of unemployment. It complements the work of Landais et al. [2010], who study optimal unemployment insurance in a similar model.

4 Fiscal Multiplier

This section specializes the model of Section 2 to capture two key properties of recessions: (i) unemployment is higher in recessions; and (ii) jobs are rationed in recessions, as there is an acute job shortage irrespective of job search and matching frictions. The government uses fiscal policy in the form of spending on public-sector jobs to reduce unemployment. This policy is widely used in practice. In the US, public employment represents 16% of aggregate employment, and Figure 1 suggests that it is countercyclical. More systematically, the correlation between public employment and unemployment rate is 0.66 in the US for the 1970–2007 period [Gomes, 2010]. And since World War II, 75% of the US government consumption of goods and services consists of public-sector wages [Cavallo, 2005]. To guide the design of fiscal policy, I study the public-employment multiplier—the reduction in unemployment achieved by spending one unit of private good on public employment—over the business cycle.

4.1 Job rationing

I introduce recessions and job rationing in the generic model of Section 2 by making two assumptions on the production function and wage schedule, as in Michaillat [forthcoming].

ASSUMPTION 1. The production function has diminishing marginal returns to labor: $\alpha < 1$.

This assumption yields a downward-sloping demand for labor in the price θ -quantity *n* diagram, which has important macroeconomic implications. This assumption is motivated by the observation that, at business cycle frequency, some production inputs are slow to adjust.



Figure 1: Share of public employment in total employment in the US, 1939–2011. *Notes:* The data are seasonally-adjusted, monthly data from the Current Employment Survey (CES) collected by the Bureau of Labor Statistics (BLS) for the 1939–2011 period. Employment in the public sector is the employment level in the government super sector. Total employment is the employment level in the total nonfarm super sector.

I use the simple wage schedule from Blanchard and Galí [2010]: $w_t = w(a_t) = \omega \cdot a_t^{\gamma}$. The parameter γ captures wage rigidity over the business cycle. If $\gamma = 0$, wages are completely fixed over the cycle. If $\gamma = 1$, wages are proportional to technology and fully flexible over the cycle.

ASSUMPTION 2. The wage schedule is rigid: $\gamma < 1$.

Wages are rigid in the sense that they adjust only partially to technology shocks. This rigidity generates unemployment fluctuations over the business cycle [Hall, 2005]. Historical, ethnographic, and empirical studies document and explain the sources of wage rigidity [Bewley, 1999; Campbell and Kamlani, 1997; Doeringer and Piore, 1971; Jacoby, 1984; Kahn, 1997; O'Brien, 1989].

4.2 Static environment

A first step in the study of the public-employment multiplier over the business cycle is to characterize the equilibrium of the model in a static environment with no aggregate shocks ($a_t = a$ for all t), a constant government policy ($b_t = b$ and $g_t = g$ for all t), and a labor market in steady state (inflows to and outflows from unemployment are equal).⁸ I focus on four labor market variables: effort e, private employment l, labor market tightness θ , and total employment n.

First, the utility-maximizing job-search effort *e* is related to labor market tightness θ by applying first-order condition (1) to a static environment:

$$[1 - \delta \cdot (1 - s)] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1 - s) \cdot k(e) = \ln\left(\frac{1}{b}\right).$$
(8)

Equation (8) implicitly defines an increasing supply of effort $e(\theta)$, since the per-unit job-finding probability $f(\theta)$ increases with θ . In steady state inflows into unemployment $s \cdot n$ equal outflows from unemployment $[1 - (1 - s) \cdot n] \cdot e(\theta) \cdot f(\theta)$, determined by unemployed workers' search effort. Hence labor supply $n^{s}(\theta)$ can be expressed as a function of labor market tightness θ :

$$n^{s}(\theta) = \frac{e(\theta) \cdot f(\theta)}{s + (1 - s) \cdot e(\theta) \cdot f(\theta)}.$$
(9)

 $n^{s}(\theta)$ is a labor supply because it gives steady-state employment when jobseekers optimally choose search effort for a given tightness θ . Labor supply $n^{s}(\theta)$ increases with θ , because (i) higher θ increases the optimal provision of effort $e(\theta)$, increasing mechanically labor supply; and (ii) higher

⁸In search-and-matching models, comparing static environments delivers the same qualitative predictions as the study of a stochastic environment [Michaillat, forthcoming; Pissarides, 2009].

 θ increases mechanically labor supply by increasing the per-unit job-finding probability $f(\theta)$.⁹

Second, the profit-maximizing private employment *l* is related to labor market tightness θ by applying first-order condition (3) to a static environment:

$$y'(l) = \frac{w}{a} + [1 - \delta \cdot (1 - s)] \frac{r}{q(\theta)}.$$
 (10)

Equation (10) implicitly defines the demand for labor in the private sector $l(\theta; a)$. Under Assumption 1, y'(l) decreases in l. Thus labor demand $l(\theta; a)$ decreases with labor market tightness θ , since the job-filling probability $q(\theta)$ decreases in θ . Intuitively when the labor market is slack, it is easy and cheap for firms to recruit, stimulating hiring. Under Assumption 2, the normalized wage $w/a = \omega \cdot a^{\gamma-1}$ decreases with technology a, so labor demand $l(\theta; a)$ increases with a. Intuitively when technology is low, wages are relatively high, depressing hiring. I define aggregate labor demand as the sum of the labor demands in the private and public sectors:

$$n^{d}(\mathbf{\theta}; a, g) = g + l(\mathbf{\theta}; a).$$
(11)

Finally, labor market tightness θ acts as a price equilibrating labor supply and labor demand:

$$n^{s}(\theta) = n^{d}(\theta; a, g) \equiv n(a, g).$$
(12)

Equation (12) implicitly defines equilibrium labor market tightness $\theta(a,g)$, and equilibrium em-

⁹The labor supply curve can be interpreted as a Beveridge curve in this model with endogenous job search because it captures the condition that inflows into and outflows from unemployment are balanced on the labor market.

ployment n(a,g). When technology *a* decreases, equilibrium labor market tightness $\theta(a,g)$ and equilibrium employment n(a,g) decrease because aggregate labor demand $n^d(\theta;a,g)$ decreases.

For a given technology *a* and public employment *g*, the labor market equilibrium can be represented by a simple labor supply-labor demand diagram [Landais et al., 2010]. The equilibrium is depicted in Figure 2 in a price θ -quantity *n* diagram. This figure plots labor demand curves for high (left panel) and low (right panel) technology. Equilibrium employment *n*(*a*, *g*) is given by the intersection of the downward-sloping labor demand curve $n^d(\theta; a, g)$ with the upward-sloping labor supply curve $n^s(\theta)$. Tightness $\theta(a, g)$ acts as a price equalizing labor supply and labor demand. If labor supply is above labor demand, a reduction in θ increases labor demand n^d by reducing recruiting costs; reduces labor supply *n*^s by reducing the per-unit job-finding probability as well as search efforts; until labor supply equals labor demand. When technology decreases in a recession, labor demand shifts inwards. The new equilibrium has higher unemployment and lower tightness.

Jobs are rationed in recessions in the sense that the labor market fails to clear and some unemployment remains even as unemployed workers exert arbitrarily large search efforts. This job shortage is depicted in Figure 2. After a negative technology shock, marginal product of labor falls but the rigid wage adjusts downwards only partially, so labor demand shifts inwards. If the adverse shock is large enough, the marginal product of the least productive workers falls below the wage. It is unprofitable for firms to hire these workers even if recruiting is costless at $\theta = 0$: labor demand cuts the x-axis at $n^R < 1$. Even if workers search infinitely hard, shifting labor supply outwards and pushing tightness θ to 0, firms never hire more than $n^R < 1$ workers: jobs are rationed.



Figure 2: Labor market equilibrium in a price θ -quantity *n* diagram

4.3 Public-employment multiplier

In a static environment parameterized by technology a and public employment g, equilibrium is described by (12). Differentiating this equilibrium condition with respect to g yields

$$\frac{\partial n^s}{\partial \theta} \cdot \frac{\partial \theta}{\partial g} = \frac{\partial n^d}{\partial \theta} \cdot \frac{\partial \theta}{\partial g} + \frac{\partial n^d}{\partial g} = \frac{\partial n}{\partial g}$$

I define the price-elasticities ε^s and ε^d of labor supply and labor demand:

$$egin{aligned} & arepsilon^s = rac{ heta}{n^s} \cdot rac{\partial n^s}{\partial heta} > 0, \ & arepsilon^d = -rac{ heta}{n^d} \cdot rac{\partial n^d}{\partial heta} > 0 \end{aligned}$$

 ε^d is normalized to be positive. The effect of public employment g on aggregate employment n is

$$\frac{\partial n}{\partial g} = \frac{\partial n^d}{\partial g} \cdot \frac{1}{1 + \left(\varepsilon^d / \varepsilon^s\right)}.$$
(13)

This formula says that the increase dn in aggregate employment following a small increase dgin public employment equals the direct increase in labor demand dn^d , attenuated by a factor $1/[1 + (\varepsilon^d/\varepsilon^s)] < 1$ that captures the reduction in labor demand due to the equilibrium increase in labor market tightness. In Figure 2, an increase dg in public employment leads to an outward shift in labor demand of amplitude dn^d . The shift in labor demand is accompanied by an increase in equilibrium labor market tightness to equilibrate labor supply with the higher level of labor demand. This increase in tightness leads to a movement along the labor demand curve that reduces employment by a factor $1/[1 + (\varepsilon^d/\varepsilon^s)] < 1$. This factor depends on the ratio of elasticities e^d/e^s because it measures the relative slope of labor demand and labor supply, which determines the reduction in employment due to the equilibrium adjustment in tightness. Finally, the publicemployment multiplier λ_g is the increase in aggregate employment achieved by spending one unit of private good on public employment. The multiplier is obtained by dividing (13) by the perperiod marginal cost of an increase in public employment $x_g = w(a) + [1 - \delta \cdot (1 - s)] \cdot r \cdot a/q(\theta)$:

$$\lambda_g = \frac{1}{x_g} \cdot \frac{\partial n^d}{\partial g} \cdot \frac{1}{1 + \left(\varepsilon^d / \varepsilon^s\right)}.$$
(14)

Studying the multiplier λ_g requires to study $\partial n/\partial g$. Labor demand (11) implies that $\partial n/\partial g = 1 + \partial l/\partial g$, so it is crucial to study the effect $\partial l/\partial g$ of public employment on private employment.

Using $\partial n^d / \partial g = 1$ as is evident from (11), formula (13) implies that

$$\frac{\partial l}{\partial g} = -\frac{1}{1 + \left(\varepsilon^s / \varepsilon^d\right)}.$$
(15)

PROPOSITION 2. $\partial l/\partial g < 0$ so $\partial n/\partial g < 1$. Under Assumption 1, $\partial l/\partial g > -1$ so $\partial n/\partial g > 0$.

Proposition 2 shows that public employment crowds out private employment: $\partial l/\partial g < 0$. To recruit new workers in the public sector the government posts more vacancies, which raises equilibrium labor market tightness. Crowding out occurs because it becomes more costly for private firms to recruit workers, so they reduce employment. Intuitively the government recruits from the same pool of unemployed workers as private firms; hiring more workers in the public sector increases competition for unemployed workers, making recruiting more expensive for private firms; as private firms face higher marginal costs of labor, they reduce employment. Critically, Proposition 2 shows that when there are diminishing marginal returns to labor, public jobs crowd out private jobs strictly less than one-for-one: $|\partial l/\partial g| < 1$. In that case public employment increases total employment: $\partial n/\partial g = 1 + \partial l/\partial g > 0$ and the public-employment multiplier λ_g is positive.

Next, I study how the multiplier λ_g fluctuates over the business cycle, assuming public employment adjusts automatically to economic conditions such that the share g/n of public jobs in the labor market remains constant. Formally, I assume that public employment g is an implicit function of technology a: $g(a) = \zeta \cdot n(a, g(a)), \zeta \in [0, 1)$. This assumption simplifies derivations because all equilibrium variables become implicit functions of a only. The central question is how crowding out $\partial l/\partial g$ varies over the business cycle. To answer this question, I compute priceelasticities ε^s and ε^d and perform comparative statics with respect to *a*. Using (8), I infer that the elasticity ε_{θ}^e of optimal search effort $e(\theta)$ with respect to labor market tightness θ is

$$\varepsilon_{\theta}^{e} = \frac{\theta}{e} \cdot \frac{\partial e}{\partial \theta} = \frac{1 - \eta}{K(a)}$$

where $K(a) = \kappa \cdot \{1 + s \cdot [\delta \cdot (1 - s)] / [1 - \delta \cdot (1 - s)] \cdot n/u\}$ is procyclical. Labor supply (9) implies

$$\varepsilon^{s} = u \cdot [\varepsilon^{e}_{\theta} + (1 - \eta)] = u \cdot (1 - \eta) \cdot \left[1 + \frac{1}{K(a)}\right].$$

Thus the price-elasticity ε^s of labor supply is countercyclical. Next, I calculate the price-elasticity ε^d of labor demand. Equation (10) implies that the elasticity ε^l_{θ} of optimal private employment $l(\theta; a)$ with respect to tightness θ is

$$\varepsilon_{\theta}^{l} = \frac{\theta}{l} \cdot \frac{\partial l}{\partial \theta} = -\frac{\eta}{1-\alpha} \cdot [1-\Omega(a)],$$

where $\Omega(a) = w(a)/mpl(a)$ characterizes the wedge between real wage w(a) and marginal product of labor $mpl(a) = a \cdot y'(l)$ in the private sector. $\Omega(a)$ is countercyclical because in recessions the rigid wage is high relative to marginal productivity. Aggregate labor demand (11) implies

$$\mathbf{\epsilon}^{d} = (1 - \zeta) \cdot \mathbf{\epsilon}_{\mathbf{\theta}}^{l} = -\eta \cdot \frac{1 - \zeta}{1 - \alpha} \cdot [1 - \Omega(a)]$$

Hence, the price-elasticity ε^d of labor demand is procyclical.

PROPOSITION 3. Assume that $g(a) = \zeta \cdot n(a, g(a))$. Under Assumptions 1 and 2:

$$\frac{\partial \left[\partial l/\partial g\right]}{\partial a} < 0.$$

Proposition 3 says that when technology *a* falls in recessions, the amplitude $|\partial l/\partial g|$ of crowding out of private jobs by public jobs falls. Hence when unemployment is high, crowding out is low and public employment reduces unemployment effectively. Intuitively when jobs are rationed in recessions, competition among employers to recruit workers is weak: in particular, competition for workers from the public sector does not hinder job creation by private firms. Without much crowding out of the private sector, a public-employment policy is effective in recessions. But in expansions, crowding out is important and public employment is ineffective at reducing unemployment. Proposition 3 implies that the public-employment multiplier λ_g is countercyclical.

COROLLARY 1. Assume that $g(a) = \zeta \cdot n(a, g(a))$. Under Assumptions 1 and 2, $d\lambda_g/da < 0$.

Corollary 1 says that when technology *a* falls in recessions, the public-employment multiplier λ_g increases. As a result, spending one dollar on public-sector jobs reduces unemployment more effectively in recessions. The result comes mostly from the decrease in the crowding-out of private employment by public employment in recessions. In addition, it becomes cheaper to hire labor in recessions as wages and recruiting costs fall.

Figure 2 illustrates the mechanism behind Proposition 3. In recessions, an outward shift in aggregate labor demand following an increase in public employment leads to a small increase in equilibrium tightness θ and thus a small decrease in private employment $l(\theta; a)$. Crowding out is

	Interpretation	Value	Source
δ	Discount factor	0.999	Corresponds to 5% annually
ζ	Share of public employment	0.16	CES, 2000–2010
S	Separation rate	0.94%	JOLTS, 2000–2010
b	UI benefit rate	60%	Pavoni and Violante [2007]
η	Effort-elasticity of matching	0.7	Petrongolo and Pissarides [2001]
γ	Real wage flexibility	0.5	Pissarides [2009], Haefke et al. [2008]
r	Recruiting cost	0.23	Barron et al. [1997], Silva and Toledo [2009]
ω_h	Effectiveness of matching	1.06	JOLTS, 2000–2010
α	Marginal returns to labor	0.67	Matches labor share of 0.66
ω	Steady-state real wage	0.71	Matches unemployment of 5.9%
к	Elasticity of disutility of effort	0.59	Matches Krueger and Mueller [2010]
ω_k	Disutility of effort	47.9	Matches search intensity of 0.085

Table 1: Parameter values in simulations (weekly frequency)

limited so the increase in aggregate employment is large. But in expansions, an outward shift in aggregate labor demand leads to a large increase in equilibrium tightness and a large decrease in private employment. Crowding out is important, the increase in aggregate employment is small.¹⁰

Collecting these results, I can provide an expression of the public-employment multiplier:

$$\lambda_g = \frac{1 - \alpha}{1 - \alpha + (1 - \zeta) \cdot \frac{\eta}{(1 - \eta)} \cdot \frac{K(a)}{1 + K(a)} \cdot \frac{1 - \Omega(a)}{u}} \cdot \frac{1}{mpl(a)}.$$
(16)

4.4 Calibration and numerical illustration

I calibrate the model to US data to quantity the cyclical fluctuations of the public-sector multiplier. I calibrate all parameters at a weekly frequency as summarized in Table 1. Following the strategy

¹⁰The crowding-out of private employment by public employment results from the competition for workers between private firms and government. In Quadrini and Trigari [2007] and Gomes [2010], the public sector influences the private sector though wages: higher wages in the public sector increase the outside option of workers, and leads through wage bargaining to higher wages in the private sector (and therefore lower private employment).

in Landais et al. [2010], I calibrate as many parameters as possible from micro-evidence for the US, and macro-data for the US for the December 2000–June 2010 period:¹¹ $\delta = 0.999$, s = 0.0094, $r = 0.32 \cdot \omega$, $\eta = 0.7$, $\gamma = 0.5$, b = 60%. I calibrate the share of public employment in total employment ζ using seasonally-adjusted, monthly data from the Current Employment Survey (CES) collected by the BLS for the 2000-2010 period. Public employment is the employment level in the government super sector. Total employment is the employment level in the total nonfarm super sector. I find that on average $\zeta = 0.16$.

I normalize steady-state technology $\overline{a} = 1$. I target the average vacancy-unemployment ratio $\overline{v}/\overline{u} = 0.47$ and unemployment rate $\overline{u} = 5.9\%$ in the US for 2000–2010. I target the conventional labor share $\overline{ls} = 0.66$. Krueger and Mueller [2010] provide evidence on job-search intensity of unemployed workers in the US, modeling job-search intensity as time allocated to job-search activities.¹² In the American Time Use Surveys (ATUS) from 2003 to 2007, unemployed workers devote 41 minutes to job search on weekdays on average. Assuming a workday of 8 hours, the fraction of time spent on job search is $41/(8 \cdot 60) = 0.085$. I set $\overline{e} = 0.085$. I target labor market tightness $\overline{\theta} = \overline{v}/(\overline{u} \cdot \overline{e}) = 0.47/0.085 = 5.53$. The last target is the elasticity of job-search effort to UI benefit rate *b*, estimated by Krueger and Mueller [2010] at -1.6. I set $\overline{e}_b^e = -1.6$.

I calibrate the six remaining parameters: production function parameters α, ω_p , utility function parameters ω_k, κ , effectiveness of matching ω_h , wage level ω , to match these steady-state

¹¹This is the longest period during which the Job Openings and Labor Turnover Survey (JOLTS), collected by the BLS, is available. This survey is critical because it contains monthly series for job openings, job separations, and new hires. As in Landais et al. [2010], I use JOLTS series for all nonfarm industries. These series are appropriate because they includes all jobs in the (nonfarm) private sector and the public sector (federal, state, and local government).

¹²Job search activities include contacting a potential employer, calling visiting an employment agency, reading and replying to job advertisements, and job interviewing.



Figure 3: Public-employment multiplier

Notes: The left panel shows the relationship between technology and unemployment in the calibrated model. The right panel represents a measure of the public-employment multiplier, computed by multiplying equation (16) by the GDP, which is $a \cdot y(l) + w \cdot g - (r \cdot a)/q(\theta) \cdot (s \cdot n)$. The graph represents the increase (in percentage point) of employment rate achieved by spending 1% of GDP on public employment. Calibration is in Table 1.

targets estimated in the data. To calibrate matching effectiveness ω_h , I use the steady-state relationship $\omega_h = s/(1-s) \cdot [(1-\overline{u})/(\overline{e} \cdot \overline{u})] \cdot \overline{\theta}^{\eta-1}$, which yields $\omega_h = 1.06$. I calibrate the production function parameter α using the steady-state relation $\overline{ls} \equiv (\overline{w} \cdot \overline{l}) / \overline{y} = \omega \cdot \overline{l}^{1-\alpha}$, which implies that the firm's first-order condition (10) is $\overline{ls} \cdot ([1-\delta \cdot (1-s)] \cdot 0.32/q(\overline{\theta})+1) = \alpha$, yielding $\alpha = 0.67$. The steady-state wage ω is related to steady-state labor share by $\omega = \overline{ls} \cdot \overline{l}^{\alpha-1}$, so $\omega = 0.71$ and recruiting cost $r = 0.32 \cdot \omega = 0.23$. In steady state, the elasticity of job search with respect to benefits (keeping labor market tightness constant) can be derived from the worker's first-order condition (8): $\overline{e}_b^e = \partial \ln(e)/\partial \ln(b) \approx (\kappa + \overline{u})/[\kappa \cdot (1+\kappa)] \cdot 1/\ln(b)$, where the approximation is valid when $\delta \approx 1$. Solving for κ , I find $\kappa = 0.59$. Finally, I rewrite (8) as $\omega_k \cdot [(1-\delta \cdot (1-s)) \cdot \{\overline{e}^{\kappa}/f(\overline{\theta})\} + \delta \cdot (1-s) \cdot \{\kappa/(\kappa+1)\} \cdot \overline{e}^{\kappa+1}] = \ln(1/b)$ to calibrate $\omega_k = 47.9$.

Figure 3 illustrates numerically the theoretical results of Corollary 1. The left panel illustrates how unemployment fluctuates with technology. Recessions correspond to periods of low technol-

ogy and high unemployment. The right panel plots the public-employment multiplier for a series of technology levels. I compute this multiplier by multiplying (16) by the Gross Domestic Product (GDP), which is $a \cdot y(l) + w \cdot g - (r \cdot a)/q(\theta) \cdot (s \cdot n)$. The graph represents the increase, measured in percentage points, of employment rate achieved by spending 1% of GDP on public employment. The cyclical fluctuations of the fiscal multiplier predicted in theory are large: the multiplier increases nearly fourfold from 0.3 to 1.1 when the unemployment rate increases from 4% to 11%.

4.5 Preliminary empirical evidence

The key theoretical result of this paper, proved in Corollary 1, is that the public-employment multiplier is countercyclical. Unfortunately empirical evidence about the behavior of fiscal multipliers over the business cycle is lacking, because studies estimating multipliers do not usually account for the state of the economy when fiscal measures are enacted [Parker, 2011]. Furthermore it is difficult to obtain precise estimates of recessionary fiscal multipliers, because few recessions occur in available data [Parker, 2011]. Finally, my theoretical results concern government spending on public-sector jobs whereas most studies lump together government spendings on goods and services, which include both spendings on public-sector jobs and spendings on goods purchases. The estimated effects of government spending on goods and services are comparable to my theoretical results only if fiscal expansions consist mostly of creation of public-sector jobs, which is plausible in the US as public-sector wages represent about 75% of government spending on goods and services and public employment is quite countercyclical [Cavallo, 2005; Gomes, 2010]. While more empirical work is required to reach a consensus about the effects of fiscal policies over the business cycle, recent studies suggest that fiscal multipliers may be countercyclical as in Corollary 1.

The most compelling evidence comes from Auerbach and Gorodnichenko [2010], who use a regime-switching structural vector autoregression (SVAR) model to capture fluctuations in fiscal multipliers over the business cycle. They find that fiscal multipliers are much larger in recessions than in expansions. Nakamura and Steinsson [2011] investigate the effect of government spending in US states during military build-ups on state employment, and find that the effect is roughly twice larger in high-unemployment than in low-unemployment periods. Last, Shoag [2011] shows that the positive effect of state government spending, instrumented using state pension-plan returns, on state unemployment is larger during times of high unemployment.¹³

Quantitatively, Figure 3 suggests that the public-employment multiplier is in the 0.3–1.1 range, and is 0.6 for an unemployment rate of 6%. Monacelli et al. [2010] use a SVAR framework to estimate the effects of fiscal policy on the US labor market for 1954–2006. They find that an increase in government spending on goods and services by 1% of GDP increases the employment rate by 0.6 percentage point at the peak, in line with my simulations. Using a SVAR with alternative identification restrictions, Pappa [2010] studies the transmission of fiscal shocks in the labor market for several developed economies. In US data for 1970–2008, she finds that the peak response to a 1% increase in government spending on public employment is a 0.15% increase in employment.¹⁴ On average, employment rate is 94% and spending on public employment is 10% of GDP in the

¹³On the other hand Canova and Pappa [forthcoming], using a SVAR framework with additional restrictions to capture recessionary periods, argue that fiscal multipliers in recessions are unlikely to be larger than those obtained on average over the business cycle. They identify recessions in the data as periods when government deficit is large and the nominal interest cannot respond to shocks (because it is at the zero lower bound).

¹⁴This result also holds at the state level in the US. Using state-level data for the US for 1969–2001, Pappa [2009] finds that for a typical state, a 1% increase in government spending on public employment leads to a 0.15% increase in employment at the peak.

US for the 1970–2008 period. Therefore the employment rate increases by 1.4 percentage points in response to an increase of spending on public employment by 1% of GDP. This public-employment multiplier of 1.4 is in line, albeit slightly above, my numerical results.¹⁵

5 Robustness

This section delimits the domain where fiscal policy can effectively reduce recessionary unemployment. First, public employment is not effective in search-and-matching models without job rationing such as (i) the canonical Pissarides [2000] model with Nash bargaining over wages; and (ii) the Hall [2005] model with rigid wages. Second, fiscal policies stimulating labor supply, such as monitoring unemployed workers' job search, have virtually no effect on unemployment when jobs are rationed in recessions. Last, other fiscal policies stimulating labor demand are effective in recessions: the *wage-subsidy multiplier*—the reduction in unemployment rate achieved by spending one unit of private good on a wage subsidy—is also positive and countercyclical.

5.1 Public employment in the absence of job rationing

Proposition 2 implies that the public-employment multiplier is positive in a model with job rationing. But fiscal policy in the form of public employment is not effective in any model of unemployment. In standard search-and-matching models, such as the Pissarides [2000] model with

¹⁵My model assumes that labor force participation does not respond to fiscal policy, such that an increase in employment translates into a reduction in unemployment. This assumption is supported by Monacelli et al. [2010], who find that fiscal policy does not affect labor force participation. Bruckner and Pappa [forthcoming] point out, however, that the response of labor market variables to fiscal policy seems to change over time in the US. For 1964–2009 they find that, in response to an increase in government spending, employment increases but unemployment also increases because of an increase in labor force participation. Their result holds in a number of OECD countries.

Nash bargaining or the Hall [2005] model with rigid wages, public jobs crowd out private jobs one-for-one and the public-employment multiplier is nil. In these models there is no job rationing. Infinitely many jobs are available, and the only hindrance to full employment is not a lack of jobs but matching frictions [Michaillat, forthcoming]. Hence public employment cannot reduce unemployment and simply replaces private jobs one-for-one with public jobs.

To capture the main features of the Pissarides [2000] model, I modify the model of Section 4.1 by assuming that (i) the production function is linear: $\alpha = 1$; and (ii) wages are determined by Nash bargaining. In a static environment parameterized by technology *a* and public employment *g*, the equilibrium $\{e(a,g), l(a,g), \theta(a,g)\}$ on the labor market is characterized by three equations: (8), (10), and (12) that can be conveniently rewritten as:

$$l+g = \frac{e \cdot f(\theta)}{s + (1-s) \cdot e \cdot f(\theta)}.$$
(17)

To determine equilibrium wage w(a,g) I solve in the Appendix the bargaining problem between worker and firm using the generalized Nash solution, which adds an extra equilibrium condition:

$$\frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right) = \left[1-\delta \cdot (1-s)\right] \cdot \frac{r}{q(\theta)} \left[1+\frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right)\right] + \delta \cdot (1-s) \cdot \frac{\kappa}{1+\kappa} \cdot e \cdot r \cdot \theta.$$
(18)

The equilibrium system of four equations {(8), (10), (17), (18)} and four variables $\{e, l, \theta, w/a\}$ is independent of technology *a*. Thus fluctuations in technology do not lead to any variations in labor market variables $\{e, l, \theta\}$. Only the equilibrium wage is fully flexible and varies in proportion to technology. Consequently there are no periods when unemployment is above average, so

there are no recessions. Furthermore, public employment has no effect on aggregate employment. Variables $\{e, \theta\}$ are fully determined by equations $\{(8), (18)\}$, independent of public employment g. Equilibrium condition (17) implies that aggregate employment n = l + g is also independent of public employment g. When public employment increases by dg, it reduces private employment by dl = -dg. Hence the public-employment multiplier $\lambda_g = 0$. On Figure 2 labor demand is horizontal because the production function is linear so labor demand (10) is fully elastic. Hence labor demand is unaffected by an increase in public employment, as is equilibrium employment.

To capture the main features of the Hall [2005] model, I modify the model of Section 4.1 by assuming that the production function is linear: $\alpha = 1$. I maintain rigid wages. This model generates large employment fluctuations but does not exhibit job rationing [Michaillat, forthcoming]. The expression (16) for the public-employment multiplier applies to this model by plugging in $\alpha = 1$. It implies that the multiplier $\lambda_g = 0$. In Figure 2, labor demand (10) is fully elastic and horizontal.

5.2 Fiscal policies stimulating labor supply

To stimulate employment, governments often try to improve matching on the labor market by monitoring unemployed workers' job search: a placement agency was implemented in Germany by the "Hartz reform" from 2003 to 2005 to advise and monitor jobseekers [Fahr and Sunde, 2009]; monitoring has been implemented in the US under the Temporary Assistance for Needy Families (TANF) program started in 1996, which specifies that unemployed recipients must participate at least to 20 hours of monitored job search per week [Pavoni and Violante, 2007]; and the "New Deal", a program for all unemployed workers introduced in the UK in 1998, includes job-search

monitoring during compulsory regular meetings with an adviser [Van Reenen, 2003].

The Appendix adds job-search monitoring to the model of Section 4.1. Unemployed workers are monitored with probability π_t , and those who are not searching when monitored receive only a fraction $z_t < 1$ of UI benefits. By increasing the monitoring probability π_t , the government increases search incentives and reduces unemployment by stimulating labor supply. In a static environment, the utility-maximizing choice of search effort *e* becomes

$$[1-\delta\cdot(1-s)]\cdot\frac{k'(e)}{f(\theta)}+\delta\cdot(1-s)\cdot\kappa\cdot k(e)=\ln(1/b)+\pi\cdot\ln(1/z)\cdot\left[\delta\cdot(1-s)+\frac{1-\delta\cdot(1-s)}{f(\theta)}\right].$$

Compared to (8), there is an additional term on the right-hand side that is positive for a monitoring probability $\pi > 0$. If $\pi > 0$, effort $e(\theta)$, labor supply $n^s(\theta)$, and equilibrium employment n(a,g) are higher. I assume that the per-period marginal cost of monitoring u unemployed workers is $x_{\pi} = u \cdot m$ in units of private good, where m is a parameter.¹⁶ In a static environment parameterized by technology a and monitoring π , the labor market equilibrium is characterized by a condition similar to (12), saying that tightness acts as a price equilibrating labor demand and labor supply:

$$n^d(\mathbf{\Theta};a) = n^s(\mathbf{\Theta};\pi) \equiv n(a,\pi).$$

I differentiate this equilibrium condition with respect to π , and proceed as in Section 4.3. The

¹⁶The resource cost of monitoring includes administrative costs, time spent interviewing jobseekers, or time spent verifying worker's search efforts.

effect of monitoring on aggregate employment is

$$\frac{\partial n}{\partial \pi} = \frac{\partial n^s}{\partial \pi} \cdot \frac{1}{1 + \left(\epsilon^s / \epsilon^d\right)}.$$
(19)

The *monitoring multiplier* λ_{π} , which is the increase in aggregate employment achieved by spending one unit of private good on monitoring, is given by

$$\lambda_{\pi} = \frac{1}{x_{\pi}} \cdot \frac{\partial n^s}{\partial \pi} \cdot \frac{1}{1 + \left(\varepsilon^s / \varepsilon^d\right)}.$$
(20)

While public employment stimulates aggregate labor demand n^d , monitoring stimulates labor supply n^s . As a consequence the monitoring multiplier (20) is increasing in the ratio of elasticities $\varepsilon^d/\varepsilon^s$, whereas the public-employment multiplier (14) is increasing in the ratio of elasticities $\varepsilon^s/\varepsilon^d$. Section 4.3 shows that ratio $\varepsilon^s/\varepsilon^d$ and public employment multiplier are countercyclical. Since the ratio $\varepsilon^d/\varepsilon^s$ is procyclical, the monitoring multiplier tends to be procyclical, and all the more so as the marginal monitoring cost x_{π} is countercyclical. The behavior of $\partial n^s/\partial \pi$ is more difficult to characterize, but the Appendix proves that the monitoring multiplier λ_{π} increases with technology when technology is below some threshold. As technology falls below this threshold, the monitoring multiplier decreases. In particular, the monitoring multiplier is low in recessions.

In recessions, increasing unemployed workers' search effort through higher monitoring has little desirable effect on unemployment because jobs are rationed and job search does not matter much. In the same way, having unemployed workers search harder by reducing the generosity of unemployment insurance has little desirable effect on unemployment in Landais et al. [2010]. On the other hand in expansions, an increase in search efforts reduces unemployment significantly. Thus the monitoring multiplier decreases when technology decreases. This theoretical result mirrors Proposition 3 in Landais et al. [2010], which proves that the macro-elasticity of unemployment with respect to net reward from work is procyclical.

To verify that the monitoring multiplier is procyclical for a plausible range of technology and unemployment, Figure 4 plots the multiplier as a function of unemployment in the calibrated model. As in Figure 3, the multiplier is multiplied by GDP to represent the increase, measured in percentage point, of employment rate achieved by spending 1% of GDP on monitoring. Cyclical fluctuations of the multiplier are large: it decreases nearly sixfold, from 1.8 to 0.3, when the unemployment rate increases from 4% to 11%. Quantitatively, this decline is similar to the decline of the macro-elasticity of unemployment plotted in Figure 3 in Landais et al. [2010]. This is because both monitoring multiplier and macro-elasticity capture the increase in aggregate employment achieved by stimulating labor supply in a model of equilibrium unemployment with job rationing.

5.3 Wage and recruiting subsidies

Empirical evidence suggests that tax cuts are expansionary [Romer and Romer, 2010]. In the model of Section 4.1 pre-tax wages are rigid so a wage subsidy, which could be implemented as a payroll tax cut, necessarily reduces unemployment. This section analyzes the effect of a wage subsidy on employment over the business cycle, captured by a wage-subsidy multiplier. This section shows that the two key findings of Section 4—a positive and countercyclical fiscal multiplier—also hold for a wage subsidy. In fact, there is a tight link between public-employment and wage-subsidy





Notes: This graph represents the monitoring multiplier computed by multiplying equation (A20) by the GDP, which is $a \cdot y(l) + w \cdot g - (r \cdot a)/q(\theta) \cdot (s \cdot n)$. The graph represents the increase (in percentage point) of employment rate achieved by spending 1% of GDP on monitoring job search. Calibration is in Table 1. In addition, I set the monitoring cost to m = 0.63 and the punishment to z = 43% (details of the calibration are in the Appendix).

multipliers because both policies stimulate aggregate labor demand.

This analysis has policy implications because payroll tax cuts on the employer's side are a common measure to stimulate employment. In the US, the Hiring Incentives to Restore Employment Act was approved in 2010 to exempt businesses from Social Security payroll taxes for newly hired worker [Kitao et al., 2010]; the American Jobs Act presented to Congress in 2011 also includes payroll tax cuts for firms from 6.2% to 3.1%, for a cost of \$70 billion.

The government provides a wage subsidy τ to reduce the marginal cost of labor. In a static

environment the profit-maximizing level of private employment l becomes

$$y'(l) = (1-\tau) \cdot \frac{w}{a} + [1-\delta \cdot (1-s)] \cdot \frac{r}{q(\theta)}.$$

The wage-subsidy multiplier λ_{τ} is the increase in aggregate employment achieved by spending one unit of private good on a wage subsidy. The per-period marginal cost of a wage subsidy is $x_{\tau} = w(a) \cdot l$. Since a wage subsidy stimulates aggregate labor demand like public employment, the methodology of Section 4.3 remains valid and the wage-subsidy multiplier is

$$\lambda_{\tau} = \frac{1}{x_{\tau}} \cdot \frac{\partial n^d}{\partial \tau} \cdot \frac{1}{1 + \left(\varepsilon^d / \varepsilon^s\right)}.$$
(21)

In the Appendix, I show that $(\partial n^d / \partial \tau) = \partial l / \partial \tau = 1/(1 - \alpha) \cdot (w(a) \cdot l) / mpl(a) = [(\partial n^d / \partial g) / x_g] \cdot x_{\tau}/(1 - \alpha)$. Comparing (21) to (14), it becomes clear that the wage-subsidy multiplier λ_{τ} is related to the public-employment multiplier λ_g by

$$\lambda_{\tau} = \frac{1}{1-\alpha} \cdot \lambda_{g}$$

Since the methodology of Section 4.3 applies to any fiscal policy stimulating labor demand, the multipliers for these policies are all very similar: differences only come from the ratio of the direct effect of the policy on labor demand (say, $\partial n^d / \partial \tau$) to the marginal cost of the policy (say, x_{τ}). These ratios are proportional for public employment and wage subsidy, so λ_{τ} and λ_g are proportional.

The cyclical fluctuations of the wage-subsidy multiplier λ_{τ} are therefore identical to those of the

public-employment multiplier λ_g . The key finding from Corollary 1 remains: the wage-subsidy multiplier is countercyclical. A wage subsidy reduces the marginal cost of labor, which leads firms to increase employment; higher aggregate employment increases labor market tightness and recruiting costs until a new equilibrium is reached, at which point the new marginal cost of labor equals the marginal product of labor. In recessions when jobs are rationed, recruiting costs are low and do not vary much with employment so a wage subsidy triggers a large increase in employment; in expansions, recruiting costs are high and increase rapidly with employment so a wage subsidy only achieves a small increase in employment; thus, a wage subsidy is more effective in recessions.

The Appendix relates the wage-subsidy multiplier λ_{τ} to a *recruiting-subsidy multiplier*, also commonly used in practice [Kluve, 2006]. I assume that the government covers a fraction $\vartheta \in [0, 1]$ of the recruiting cost $r \cdot a/q(\theta)$ paid by firms for each new hire. The relation between wagesubsidy and recruiting-subsidy multipliers is very simple: $\lambda_{\vartheta} = [1 - \delta \cdot (1 - s)]/s \cdot \lambda_{\tau}$. If $\delta \approx 1$, then $\lambda_{\vartheta} \approx \lambda_{\tau}$. These subsidies are broadly equivalent. They have the same effect on employment because they both consist in a transfer to firms for each marginal hire. The way the transfer is paid (through wages or recruiting costs) is irrelevant: what matters is that firms face a lower marginal cost of labor. The recruiting-subsidy multiplier is obviously countercyclical.

6 Towards the Design of Optimal Fiscal Policies

I conclude by discussing the implications of the paper for the design of fiscal policies, as well as the limitations to overcome to characterize fully welfare-maximizing fiscal policies. Designing optimal fiscal policies requires joint empirical and theoretical efforts to measure the effectiveness of fiscal policies in the data, and design fiscal policies maximizing social welfare subject to the government's budget constraint. This section is articulated along the empirical and theoretical challenges to designing optimal fiscal policies.

6.1 Estimating fiscal multipliers

On the empirical front, a critical implication of this paper is that the empirical estimates of fiscal multipliers reported in the literature, which are averages over all possible states of the economy, do not apply in recessions because government spending and tax multipliers vary widely over the business cycle.¹⁷ Since recessionary multipliers are very different from average multipliers, average multipliers fail to characterize the effectiveness of a fiscal expansion in recessions. They do not convey much useful information to policymakers. Future studies should aim to estimate multipliers over the business cycle instead of average multipliers. As pointed out by Parker [2011], estimating state-dependent multipliers may require to upgrade our empirical methods to move beyond the linear dynamics characterized by VAR and linearized DSGE models.

¹⁷Ramey [2011] provides a comprehensive survey of this literature. There is also a large literature in labor economics that estimates the effectiveness of active labor market policies, designed to help unemployed workers find a new job [Card et al., 2009; Heckman et al., 1999]. These papers study microeconomic instead of macroeconomic outcomes: they estimate how much the unemployment duration of a treated unemployment worker decreases, and not how much local unemployment falls when a program is implemented. Hence these results are not directly relevant for the design of macroeconomic policy: as pointed out by Landais et al. [2010] in the context of unemployment insurance, these micro-estimates may differ greatly from macro-estimates because of general-equilibrium effects arising in a labor market with job rationing.

6.2 Characterizing optimal fiscal policies

On the theoretical front, this paper proposes an equilibrium unemployment framework in which jobs are rationed in recessions. The framework can be used to study the effects of government spending and taxes on unemployment over the business cycle. The cyclical behavior of multipliers suggests that a desirable fiscal policy likely stimulates aggregate labor demand in recessions, for instance through public employment. However, the multipliers provide only a partial picture of the welfare effects of fiscal policies. Characterizing optimal fiscal policy requires to maximize social welfare subject to the government's budget constraint. This paper is a first attempt at studying fiscal policy in recessions theoretically, and my simple model should be extended to study welfare-maximizing fiscal policies over the business cycle.¹⁸

First, the only source of recessions in the model are technology shocks. Future work should explore how other shocks, in particular aggregate demand shocks or financial disturbances, influence the effectiveness of fiscal policies. For instance if firms were constrained by a low and inelastic goods demand in recessions, then a wage subsidy would be completely ineffective.

Second, the rigid wage schedule specified in the model, while theoretically and empirically

¹⁸A large literature studies welfare-maximizing labor market policies in search-and-matching models, but there are no aggregate shocks and no business cycles in these models. Boone et al. [2007] solve for optimal unemployment insurance when job-search effort is unobservable, and examine whether the optimal policy involves monitoring of search effort and benefit sanctions. Kitao et al. [2010] analyze the effects of various hiring subsidies to stimulate hiring on equilibrium unemployment in the model of Mortensen and Pissarides [1994]. Mortensen and Pissarides [2002] investigate the effects of a linear payroll tax, a job destruction tax and unemployment compensation on labor market variables and characterize the optimal policy, which corrects search externalities. Hungerbühler et al. [2006] study the optimal tax policy in a search-and-matching model with a continuum of unobservable productivity levels. Gomes [2010] studies optimal public-employment policy in the Pissarides [2000] model. Gomes [2010] does examine the properties of optimal public employment when the economy is subject to technology shocks. However as acknowledged by the author, there are no recessions in his model: after a 1% negative productivity shock, the unemployment rate increases at most by 0.05 percentage points, 10 times less than in the data. This lack of amplification in a search-and-matching model with Nash bargaining was first pointed out by Shimer [2005]: this model is unable to generate enough fluctuations in unemployment in response to technology shocks because bargained wages are too flexible.

valid, does not explain where wage rigidity comes from. It is critical to design a micro-founded wage-setting mechanism to understand the effects of labor market policies on employment through wages. For instance this mechanism could explain the influence of a wage subsidy on the after-tax wage paid by employers, or the influence of public-sector wages on private-sector wages. Another issue to resolve is whether real or nominal wages are rigid. The econometric studies used to calibrate the model exploit micro-data for wages received by existing workers and new hires to show that real wages are somewhat rigid over the business cycle [Haefke et al., 2008; Martins et al., 2010; Pissarides, 2009]. But ethnographic and econometric evidence suggests that nominal wages may be more rigid than real wages [Bewley, 1999; Campbell and Kamlani, 1997; Card and Hyslop, 1996; Kahn, 1997]. If that is the case, monetary policy could be used instead of fiscal policy to tackle unemployment. If wages exhibit some nominal rigidity, some inflation would erode real wages and stimulate employment [Akerlof et al., 1996].

Finally to characterize the welfare-maximizing amount of employment in the public sector, it is necessary to quantify the welfare effects of government production. A large literature studies the impact of government consumption and government capital on economic outcomes and social welfare, with mixed results [Aschauer, 1989; Evans and Karras, 1994; Garcia-Milà and McGuire, 1992; Holtz-Eakin, 1993; Ratner, 1983]. More work is needed in this direction, for instance to assess the interaction of government capital with private production through better infrastructure, better institution, or better education.

References

- Abraham, Katharine G. and Lawrence F. Katz, "Cyclical Unemployment: Sectoral Shifts or Aggregate Disturbances?," *Journal of Political Economy*, 1986, *94* (3), 507–522.
- Aiyagari, S. Rao, Lawrence J. Christiano, and Martin Eichenbaum, "The Output, Employment, and Interest Rate Effects of Government Consumption," *Journal of Monetary Economics*, 1992, *30* (1), 73–86.
- Akerlof, George A., William T. Dickens, and George L. Perry, "The Macroeconomics of Low Inflation," Brookings Papers on Economic Activity, 1996, 1996 (1), 1–76.
- Andolfatto, David, "Business Cycles and Labor-Market Search," *American Economic Review*, 1996, 86 (1), 112–132.
- Aschauer, David A., "Is Public Expenditure Productive?," *Journal of Monetary Economics*, 1989, 23 (2), 177–200.
- Auerbach, Alan J. and Yuriy Gorodnichenko, "Measuring the Output Responses to Fiscal Policy," Working Paper 16311, National Bureau of Economic Research 2010.
- Barro, Robert J., "Long-Term Contracting, Sticky Prices, and Monetary Policy," *Journal of Monetary Economics*, 1977, *3* (3), 305–316.
- Barron, John M., Mark C. Berger, and Dan A. Black, "Employer Search, Training, and Vacancy Duration," *Economic Inquiry*, 1997, 35 (1), 167–92.
- Baxter, Marianne and Robert G. King, "Fiscal Policy in General Equilibrium," *American Economic Review*, 1993, 83 (3), 315–334.
- Bewley, Truman F., *Why Wages Don't Fall During a Recession*, Cambridge, MA: Harvard University Press, 1999.
- Blanchard, Olivier J. and Jordi Galí, "Labor Markets and Monetary Policy: A New-Keynesian Model with Unemployment," *American Economic Journal: Macroeconomics*, 2010, 2 (2), 1–30.
- and Peter Diamond, "The Beveridge Curve," *Brookings Papers on Economic Activity*, 1989, 1989 (1), 1–76.
- Boone, Jan, Peter Fredriksson, Bertil Holmlund, and Jan C. Van Ours, "Optimal Unemployment Insurance with Monitoring and Sanctions," *The Economic Journal*, 2007, *117* (518), 399–421.
- Brenner, M. Harvey and Anne Mooney, "Unemployment and Health in the Context of Economic Change," Social Science & Medicine, 1983, 17 (16), 1125–1138.
- Bruckner, Markus and Evi Pappa, "Fiscal Expansions, Unemployment and Labor Force Participation: Theory and Evidence," *International Economic Review*, forthcoming.
- Campbell, Carl M. and Kunal S. Kamlani, "The Reasons for Wage Rigidity: Evidence From a Survey of Firms," *Quarterly Journal of Economics*, 1997, *112* (3), 759–789.
- Canova, Fabio and Evi Pappa, "Fiscal Policy, Pricing Frictions and Monetary Accommodation," *Economic Policy*, forthcoming.
- Card, David and Dean Hyslop, "Does Inflation "Grease the Wheels of the Labor Market"?," Working Paper 5538, National Bureau of Economic Research April 1996.
- ____, Jochen Kluve, and Andrea Weber, "Active Labor Market Policy Evaluations: A Meta-Analysis," Discussion Paper Series 4002, IZA 02 2009.
- Cavallo, Michele, "Government Employment Expenditure and the Effects of Fiscal Policy Shocks," Working

Paper 2005-16, Federal Reserve Bank of San Francisco September 2005.

- Christiano, Lawrence J. and Martin Eichenbaum, "Current Real-Business-Cycle Theories and Aggregate Labor-Market Fluctuations," *American Economic Review*, 1992, 82 (3), 430–450.
- Christiano, Lawrence, Martin Eichenbaum, and Sergio Rebelo, "When Is the Government Spending Multiplier Large?," *Journal of Political Economy*, 2011, *119* (1), 78–121.
- Doeringer, Peter B. and Michael J. Piore, *Internal Labor Markets and Manpower Analysis*, Lexington, MA: Heath Lexington Books, 1971.
- Evans, Paul and Georgios Karras, "Are Government Activities Productive? Evidence from a Panel of U.S. States," *Review of Economics and Statistics*, 1994, 76 (1), 1–11.
- Fahr, René and Uwe Sunde, "Did the Hartz Reforms Speed-Up the Matching Process? A Macro- Evaluation Using Empirical Matching Functions," *German Economic Review*, 2009, *10* (3), 284–316.
- Finn, Mary G., "Cyclical Effects of Government's Employment and Goods Purchases," International Economic Review, 1998, 39 (3), 635–657.
- Fishback, Price V., Shawn Kantor, and John Joseph Wallis, "Can the New Deal's three Rs be rehabilitated? A program-by-program, county-by-county analysis," *Explorations in Economic History*, 2003, 40 (3), 278–307.
- Fleck, Robert K., "The Marginal Effect of New Deal Relief Work on County-Level Unemployment Statistics," *Journal of Economic History*, 1999, 59 (3), 659–687.
- Galí, Jordi, J. David Lopez-Salido, and Javier Valles, "Understanding the effects of government spending on consumption," *Journal of the European Economic Association*, 2007, 5 (1), 227–270.
- Garcia-Milà, Teresa and Therese J. McGuire, "The Contribution of Publicly Provided Inputs to States" Economies," *Regional Science and Urban Economics*, 1992, 22 (2), 229–241.
- Gertler, Mark, Luca Sala, and Antonella Trigari, "An Estimated Monetary DSGE Model with Unemployment and Staggered Nominal Wage Bargaining," *Journal of Money, Credit and Banking*, 2008, 40 (8), 1713–1764.
- Gomes, Pedro, "Fiscal Policy and the Labour Market: The Effects of Public Sector Employment and Wages," Discussion Paper 5321, Institute for the Study of Labor (IZA) 2010.
- Grubb, David, "Assessing The Impact Of Recent Unemployment Insurance Extensions In The United States," 2011.
- Haefke, Christian, Marcus Sonntag, and Thijs Van Rens, "Wage Rigidity and Job Creation," Discussion Paper 3714, Institute for the Study of Labor (IZA) 2008.
- Hall, Robert E., "Employment Fluctuations with Equilibrium Wage Stickiness," American Economic Review, 2005, 95 (1), 50–65.
- Hamilton, Gayle, Stephen Freedman, Lisa Gennetian, Charles Michalopoulos, Johanna Walter, Diana Adams-Ciardullo, Anna Gassman-Pines, Sharon McGroder, Martha Zaslow, Surjeet Ahluwalia, Jennifer Brooks, Electra Small, and Bryan Ricchett, "How Effective Are Different Welfare-to-Work Approaches? Five-Year Adult and Child Impacts for Eleven Programs," National Evaluation of Welfare-to-Work Strategies, U.S. Department of Health and Human Services 2001.
- Hawton, Keith and Stephen Platt, "Suicidal Behaviour and the Labour Market," in Keith Hawton and Kees van Heeringen, eds., *The International Handbook of Suicide and Attempted Suicide*, West Sussex, England: John Wiley & Sons, Ltd, 2000, pp. 309–384.

- Heckman, James J., Robert J. LaLonde, and Jeffrey A. Smith, "The Economics and Econometrics of Active Labor Market Programs," in Orley Ashenfelter and David Card, eds., *Handbooks in Labor Economics*, Vol. 5 1999, pp. 1865–2095.
- Holmlund, Bertil and Johan Linden, "Job Matching, Temporary Public Employment, and Equilibrium Unemployment," *Journal of Public Economics*, 1993, *51* (3), 329–343.
- Holtz-Eakin, Douglas, "State-specific estimates of state and local government capital," *Regional Science* and Urban Economics, 1993, 23 (2), 185–209.
- Hörner, Johannes, L. Rachel Ngai, and Claudia Olivetti, "Public Enterprises And Labor Market Performance," *International Economic Review*, 2007, 48 (2), 363–384.
- Hungerbühler, Mathias, Etienne Lehmann, Alexis Parmentier, and Bruno Van Der Linden, "Optimal Redistributive Taxation in a Search Equilibrium Model," *Review of Economic Studies*, 2006, 73 (3), 743 –767.
- Jacoby, Sanford, "The Development of Internal Labor Markets in American Manufacturing Firms," in Paul Osterman, ed., *Internal Labor Markets*, Cambridge, MA: MIT Press, 1984.
- Kahn, Shulamit, "Evidence of Nominal Wage Stickiness from Microdata," *American Economic Review*, 1997, 87 (5), 993–1008.
- Kitao, Sagiri, Aysegul Sahin, and Joseph Song, "Subsidizing Job Creation in the Great Recession," Staff Report 451, Federal Reserve Bank of New York 2010.
- Kluve, Jochen, "The Effectiveness of European Active Labor Market Policy," Discussion Paper 2018, Institute for the Study of Labor (IZA) 2006.
- Krueger, Alan B. and Andreas Mueller, "Job Search and Unemployment Insurance: New Evidence From Time Use Data," *Journal of Public Economics*, 2010, *94* (3-4), 298–307.
- Landais, Camille, Pascal Michaillat, and Emmanuel Saez, "Optimal Unemployment Insurance over the Business Cycle," Working Paper 16526, National Bureau of Economic Research 2010.
- Leeper, Eric M., Todd B. Walker, and Shu-Chun S. Yang, "Government Investment and Fiscal Stimulus," *Journal of Monetary Economics*, 2010, 57 (8), 1000–1012.
- Martin, John P. and David Grubb, "What Works and for Whom: A Review of OECD Countries' Experiences with Active Labour Market Policies," *Swedish Economic Policy Review*, 2001, 8 (2), 9–56.
- Martins, Pedro S., Gary Solon, and Jonathan Thomas, "Measuring What Employers Really Do about Entry Wages over the Business Cycle," Working Paper 15767, National Bureau of Economic Research 2010.
- Merz, Monika, "Search in the Labor Market and the Real Business Cycle," *Journal of Monetary Economics*, 1995, *36* (2), 269–300.
- Michaillat, Pascal, "Do Matching Frictions Explain Unemployment? Not in Bad Times.," American Economic Review, forthcoming.
- Monacelli, Tommaso, Roberto Perotti, and Antonella Trigari, "Unemployment Fiscal Multipliers," *Journal* of Monetary Economics, 2010, 57 (5), 531–553.
- Mortensen, Dale T. and Christopher A. Pissarides, "Job Creation and Job Destruction in the Theory of Unemployment," *Review of Economic Studies*, 1994, *61* (3), 397–415.
- __ and __, "Taxes, Subsidies and Equilibrium Labor Market Outcomes," Discussion Paper 0519, Center for Economic Performance 2002.
- Nakamura, Emi and Jon Steinsson, "Fiscal Stimulus in a Monetary Union: Evidence from U.S. Regions,"

2011.

- O'Brien, Anthony Patrick, "A Behavioral Explanation for Nominal Wage Rigidity During the Great Depression," *Quarterly Journal of Economics*, 1989, *104* (4), 719–735.
- Pappa, Evi, "The Effects of Fiscal Shocks on Employment and the Real Wage," *International Economic Review*, 2009, *50* (1), 217–244.
- _, "Government Spending Multipliers: An International Comparison," 2010.
- Parker, Jonathan A, "On Measuring the Effects of Fiscal Policy in Recessions," *Journal of Economic Literature*, 2011, 49, 703–718.
- Pavoni, Nicola and Giovanni L. Violante, "Optimal Welfare-To-Work Programs," *Review of Economic Stud*ies, 2007, 74 (1), 283–318.
- Petrongolo, Barbara and Christopher A. Pissarides, "Looking into the Black Box: A Survey of the Matching Function," *Journal of Economic Literature*, 2001, *39* (2), 390–431.
- Pissarides, Christopher A., Equilibrium Unemployment Theory, 2nd ed., Cambridge, MA: MIT Press, 2000.
- __, "The Unemployment Volatility Puzzle: Is Wage Stickiness the Answer?," *Econometrica*, 2009, 77 (5), 1339–1369.
- Quadrini, Vincenzo and Antonella Trigari, "Public Employment and the Business Cycle," *Scandinavian Journal of Economics*, 2007, *109* (4), 723–742.
- Ramey, Valerie A., "Can Government Purchases Stimulate the Economy?," *Journal of Economic Literature*, 2011, 49, 673–685.
- Ratner, Jonathan B., "Government Capital And The Production Function For U.S. Private Output," *Economics Letters*, 1983, *13* (2-3), 213–217.
- Romer, Christina D. and David Romer, "The Macroeconomic Effects of Tax Changes: Estimates Based on a New Measure of Fiscal Shocks," *American Economic Review*, 2010, *100* (3), 763–801.
- _ and Jared Bernstein, "The Job Impact of the American Recovery and Reinvestment Plan.," 2009.
- Shimer, Robert, "The Cyclical Behavior of Equilibrium Unemployment and Vacancies," American Economic Review, 2005, 95 (1), 25–49.
- Shoag, Daniel, "The Impact of Government Spending Shocks: Evidence on the Multiplier from State Pension Plan Returns," 2011.
- Silva, José I. and Manuel Toledo, "Labor Turnover Costs and the Behavior of Vacancies and Unemployment," *Macroeconomic Dynamics*, 2009, *13* (1), 76–96.
- Sullivan, Daniel G. and Till M. von Wachter, "Job Displacement and Mortality: An Analysis using Administrative Data," *Quarterly Journal of Economics*, 2009, *124* (3), 1265–1306.
- Van Reenen, John, "Active Labour Market Policies and the British New Deal for the Young Unemployed in Context," Working Paper 9576, National Bureau of Economic Research 2003.
- von Wachter, Till M., Jae Song, and Joyce Manchester, "Long-Term Earnings Losses due to Mass Layoffs During the 1982 Recession: An Analysis Using U.S. Administrative Data from 1974 to 2004," 2009.
- Woodford, Michael, "Simple Analytics of the Government Expenditure Multiplier," *American Economic Journal: Macroeconomics*, 2011, *3*, 1–35.

Appendix – NOT FOR PUBLICATION

A Proofs

A.1 Notations

I assume that independent of technology *a*, the government follows constant policies. So the UI benefit rate *b* remain constant, and the ratio of public to aggregate employment $\zeta = g/n$. Accordingly *a* fully characterize the state of the economy in a static environment. I define the following functions:

• Labor supply: $n^{s}(e, \theta)$ is increasing in *e* and θ , and is defined by

$$n^{s}(e, \theta) = \frac{e \cdot f(\theta)}{s + (1 - s) \cdot e \cdot f(\theta)}.$$

• Labor demand: for $\alpha < 1$, $l^d(\theta, a)$ is increasing in a, decreasing in θ , and is defined by

$$l^{d}(\theta, a) = \left[\frac{1}{lpha}\left(\omega \cdot a^{\gamma-1} + \frac{r}{q(\theta)}\right)\right]^{1/(\alpha-1)}.$$

• Effort supply: $e^{s}(\theta, b)$ is increasing in θ and Δv , and is defined implicitly by

$$[1 - \delta \cdot (1 - s)] \cdot \frac{k'(e^s)}{f(\theta)} + \kappa \cdot \delta \cdot (1 - s) \cdot k(e^s) = \ln\left(\frac{1}{b}\right).$$

• Equilibrium labor market tightness: $\theta(a)$ is defined implicitly by

$$\frac{1}{1-\zeta} \cdot l^d(\theta, a) = n^s \left(e^s(\theta), \theta \right).$$
(A1)

• Equilibrium effort: e(a) is defined by

$$e(a) = e^s(\theta(a)).$$

• Equilibrium private employment: l(a) is defined by

$$l(a) = l^d(\theta(a), a).$$

• Equilibrium public employment: g(a) is defined by

$$g(a) = \frac{\zeta}{1-\zeta} \cdot l(a).$$

• Equilibrium aggregate employment: n(a) is defined by

$$n(a) = n^{s}(e(a), \theta(a)).$$

• Equilibrium unemployment: u(a) is defined by

$$u(a) = 1 - (1 - s) \cdot n(a).$$

• Equilibrium marginal product of labor: mpl(a) is defined by

$$mpl(a) = a \cdot y'(l(a)) = w(a) + [1 - \delta \cdot (1 - s)] \cdot \frac{r \cdot a}{q(\theta(a))}.$$

• Wedge between wage and marginal product of labor: $\Omega(a)$ is defined by

$$\Omega(a) == \frac{w(a)}{mpl(a)} = 1 - [1 - \delta \cdot (1 - s)] \cdot \frac{r}{q(\theta(a))} \cdot l(a)^{1 - \alpha}, \tag{A2}$$

• A measure of the elasticity of job-search effort to labor market condition: K(a) is defined by

$$K(a) = \kappa \cdot \left[1 + s \cdot \frac{\delta \cdot (1 - s)}{1 - \delta \cdot (1 - s)} \cdot \frac{n(a)}{u(a)} \right].$$
(A3)

A.2 Some preliminary comparative-static results

LEMMA A1. *If* $\gamma \in [0, 1)$ *and* $\alpha \in (0, 1)$ *, equilibrium variables satisfy:*

$$\frac{\partial \theta}{\partial a} > 0, \ \frac{\partial e}{\partial a} > 0, \ \frac{\partial n}{\partial a} > 0, \ \frac{\partial n}{\partial a} > 0, \ \frac{\partial l}{\partial a} > 0, \ \frac{\partial u}{\partial a} < 0.$$

Proof. Effort and labor supply satisfy:

$$\frac{de^s}{d\theta} > 0, \ \frac{\partial n^s}{\partial \theta} > 0, \ \frac{\partial n^s}{\partial e} > 0.$$
 (A4)

If $\gamma \in [0,1)$ and $\alpha \in (0,1)$, then labor demand satisfy:

$$\frac{\partial l^d}{\partial \theta} < 0, \frac{\partial l^d}{\partial a} > 0$$

Differentiating equilibrium condition (A1) with respect to a yields:

$$(1-\zeta) \cdot \left[\frac{\partial n^{s}}{\partial e} \cdot \frac{\partial e^{s}}{\partial \theta} + \frac{\partial n^{s}}{\partial \theta}\right] \cdot \frac{\partial \theta}{\partial a} = \frac{\partial l^{d}}{\partial a} + \frac{\partial l^{d}}{\partial \theta} \cdot \frac{\partial \theta}{\partial a}$$
$$\frac{\partial \theta}{\partial a} = \underbrace{\frac{\partial l^{d}}{\partial a}}_{+} \cdot \left[(1-\zeta) \cdot \left\{\underbrace{\frac{\partial n^{s}}{\partial e}}_{+} \cdot \frac{\partial e^{s}}{\partial \theta} + \underbrace{\frac{\partial n^{s}}{\partial \theta}}_{+}\right\} - \underbrace{\frac{\partial l^{d}}{\partial \theta}}_{-}\right]^{-1}.$$

Thus $\frac{\partial \theta}{\partial a} > 0$. I conclude by using the comparative statics (A4) and noting that $e(a) = e^s(\theta(a))$, $n(a) = n^s(e(a), \theta(a))$, $u(a) = 1 - (1 - s) \cdot n(a)$, and $l(a) = (1 - \zeta) \cdot n(a)$.

A.3 Proof of Proposition 1

The proof first derives the first-order necessary conditions for the efficient allocation. It then follows the argument highlighted in Section 3.2. Since workers are risk-averse, the efficient allocation provides full insurance against unemployment: $c_t^l = c_t^g = c_t^u = c_t$. Accordingly, the Lagrangian of the government's problem is

$$\begin{split} L &= \sum_{t=0}^{+\infty} \delta^{t} \cdot \left\{ \chi \cdot \ln(a_{t} \cdot \omega_{p} \cdot y(g_{t})) + \ln(c_{t}) - [1 - (1 - s)n_{t-1}] \cdot k(e_{t}) \right. \\ &+ \mu_{t}^{1} \cdot \{a_{t} \cdot y(l_{t}) - c_{t} - \frac{r \cdot a}{q(\theta_{t})} \cdot \{n_{t} - (1 - s) \cdot n_{t-1}\} \\ &+ \mu_{t}^{2} \cdot \{(1 - s) \cdot (1 - e_{t} \cdot f(\theta_{t})) \cdot n_{t-1} + e_{t} \cdot f(\theta_{t}) - n_{t}\} \\ &+ \mu_{t}^{3} \cdot \{n_{t} - l_{t} + g_{t}\} \right\}. \end{split}$$

where $\{\mu_t^1, \mu_t^2, \mu_t^3\}_{t=0}^{+\infty}$ are Lagrange multipliers. The first-order conditions with respect to $\{c_t, e_t, \theta_t, n_t, l_t, g_t\}_{t=0}^{+\infty}$ yield respectively

$$1/c_t = \mu_t^1 \tag{A5}$$

$$u_t \cdot k'(e_t) = \mu_t^2 \cdot u_t \cdot f(\theta_t) \tag{A6}$$

$$\mu_t^2 \cdot e_t \cdot q(\theta_t) \cdot u_t = r \cdot a \cdot \frac{\eta}{1 - \eta} \cdot \frac{1}{f(\theta_t)} \cdot \mu_t^1 \cdot h_t \tag{A7}$$

$$\mu_t^1 \cdot \frac{r \cdot a_t}{q(\theta_t)} + \mu_t^2 = \mu_t^3 + \delta \cdot (1 - s) \cdot \mathbb{E}_t \left[\mu_{t+1}^1 \cdot \frac{r \cdot a_{t+1}}{q(\theta_{t+1})} + \mu_{t+1}^2 \cdot [1 - e_{t+1} \cdot f(\theta_{t+1})] + k(e_{t+1}) \right]$$
(A8)
$$\mu_t^3 = \mu_t^1 \cdot a_t \cdot v'(l_t)$$
(A9)

$$\mu_t^3 = \chi \cdot a_t \cdot \omega_p \cdot \frac{y'(g_t)}{p_t}.$$
(A10)

where I simplify the notations by using

$$u_t = 1 - (1 - s) \cdot n_{t-1}$$

$$h_t = n_t - (1 - s) \cdot n_{t-1}.$$

Noting that $e_t \cdot f(\theta_t) = h_t/u_t$, the first-order condition (A7) with respect to θ_t yields

$$\frac{\mu_t^2}{\mu_t^1} = \frac{\eta}{1-\eta} \cdot \frac{r \cdot a_t}{q(\theta_t)}.$$
(A11)

Combining (A11) with the first-order condition (A5) with respect to c_t and the first-order condition (A6) with respect to e_t yields

$$k'(e_t) = \frac{\eta}{1-\eta} \cdot r \cdot \theta_t \cdot \frac{1}{c_t/a_t},$$

which determines optimal job-search effort e_t . We assume that ω_k is large enough such that there is an interior solution for the optimal effort and $e_t < 1$.¹⁹ Combining the first-order conditions (A9) and (A10) with respect to l_t and g_t respectively yields

$$\chi \cdot \omega_p \cdot \frac{c_t/a_t}{p_t/a_t} = \frac{y'(l_t)}{y'(g_t)},$$

which determines optimal public employment g_t . Combining the first-order conditions (A8) and (A9) with respect to n_t and l_t respectively, dividing by $a_t \cdot \mu_t^1$, yields:

$$y'(l_t) = \frac{r}{q(\theta_t)} + \frac{\mu_t^2}{\mu_t^1 a_t} - \delta(1-s)\mathbb{E}_t \left[\frac{\mu_{t+1}^1}{\mu_t^1 a_t} \left[\frac{ra_{t+1}}{q(\theta_{t+1})} + \frac{\mu_{t+1}^2}{\mu_{t+1}^1} \left[1 - e_{t+1}f(\theta_{t+1}) \right] + \frac{k(e_{t+1})}{\mu_{t+1}^1} \right] \right]$$
(A12)

Using the isoelasticity of the disutility of effort and equation (A11):

$$k(e_t) = e_t \cdot k'(e_t) \cdot \frac{1}{\kappa + 1}$$

$$k(e_t) = e_t \cdot f(\theta_t) \cdot \frac{1}{\kappa + 1} \cdot \mu_t^2.$$
(A13)

¹⁹The optimal effort $e_t < 1$ if

$$\omega_k > \frac{\eta}{1-\eta} \cdot \frac{1}{c_t/a_t} \cdot r \cdot \theta_t$$

The right-hand side of this inequality is a constant because, as we will see, the efficient allocation is constant over time, independent of the technology process $\{a_t\}$.

Combining (A11), (A12), and (A13), and multiplying by $(1 - \eta)$, yields

$$(1-\eta)\cdot y'(l_t) = \frac{r}{q(\theta_t)} - \delta \cdot (1-s) \cdot \mathbb{E}_t \left[\frac{c_t/a_t}{c_{t+1}/a_{t+1}} \cdot \left\{ \frac{r}{q(\theta_{t+1})} - \eta \cdot r \cdot \frac{\kappa}{1+\kappa} \cdot e_{t+1} \cdot \theta_{t+1} \right\} \right]$$

This last equation determines optimal private employment l_t , and completes the characterization of the efficient allocation in a stochastic environment.

These 3 first-order conditions, together with the production constraints (2) and (5), the resource constraint (6), and the law of motion of employment (7), constitute the system of 7 equations that characterize the efficient allocations. As explained in Section 3.2, this system does not involve the technology stochastic process $\{a_t\}_{t=0}^{+\infty}$. If the efficient allocation (the solution to this system of 7 equations) is unique, then the efficient allocation is independent of the realizations of the stochastic process for technology: the efficient allocation is deterministic. If initial employment l_{-1} and g_{-1} are chosen adequately to be at their steady-state values, then the labor market variables in the efficient allocation remain constant over time.

A.4 **Proof of Proposition 2**

In a static environment parameterized by technology *a*, the equilibrium $\{\theta(a), e(a), l(a), g(a)\}$ on the labor market is characterized in Section A.1. I now consider a marginal change *dg* in public employment. The remaining equilibrium variables $\{\theta, e, l\}$ evolve according to the system of three equations: (17), (8), and (10). I report these three equations here for convenience:

$$l+g = \frac{e \cdot f(\theta)}{s + (1-s) \cdot e \cdot f(\theta)}$$
$$\ln(1/b) = [1 - \delta \cdot (1-s)] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1-s) \cdot k(e)$$
$$y'(l) = \frac{w}{a} + [1 - \delta \cdot (1-s)] \frac{r}{q(\theta)}.$$

I log-linearize this system to obtain the effect of a marginal change in dg in public employment. I start by log-linearizing the worker's optimality condition:

$$\begin{split} \ln(1/b) &= \left[1 - \delta \cdot (1 - s)\right] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1 - s) \cdot k(e) \\ 0 &= (1 - \delta \cdot (1 - s)) \cdot \left[\frac{k'(e)}{f(\theta)}\right] \left[\kappa \check{e} - (1 - \eta) \cdot \check{\theta}\right] + \kappa \cdot \delta \cdot (1 - s) \cdot k(e) \cdot [1 + \kappa] \check{e} \\ 0 &= \left[\kappa \check{e} - (1 - \eta) \cdot \check{\theta}\right] + \check{e} \cdot \kappa \cdot \frac{\delta \cdot (1 - s)}{1 - \delta \cdot (1 - s)} \cdot f(\theta) \frac{(1 + \kappa)k(e)}{k'(e)} \\ (1 - \eta) \cdot \check{\theta} &= \kappa \check{e} + \check{e} \cdot \kappa \cdot \frac{\delta \cdot (1 - s)}{1 - \delta \cdot (1 - s)} \cdot f(\theta) \cdot e \\ (1 - \eta) \cdot \check{\theta} &= \kappa \check{e} \left[1 + \frac{\delta \cdot (1 - s)}{1 - \delta \cdot (1 - s)} \cdot f(\theta) \cdot e\right] \end{split}$$

Hence, the log-linear system becomes:

$$(1-\zeta) \cdot \check{l} + \zeta \cdot \check{g} = u(a) \cdot \left[\check{e} + (1-\eta) \cdot \check{\theta}\right]$$
(A14)

$$K(a) \cdot \check{e} = (1 - \eta) \cdot \theta \tag{A15}$$

$$(\alpha - 1) \cdot \check{l} = \eta \cdot [1 - \Omega(a)] \cdot \check{\theta}$$

where $\Omega(a)$ characterizes the wedge between real wage w(a) and marginal product of labor $mpl(a) = a \cdot y'(l)$ in the private sector, and is defined by (A2) and K(a) characterizes the elasticity of jobsearch effort with respect to labor market conditions and is defined by (A3).

To derive the public-employment multiplier, it is convenient to rewrite the log-linear system as:

$$\begin{split} \check{e} &= (1-\eta) \cdot \frac{1}{K(a)} \cdot \check{\theta} \\ \check{\theta} &= -\frac{(1-\alpha)}{\eta} \cdot \frac{1}{1-\Omega(a)} \cdot \check{l} \\ \zeta \cdot \check{g} &= -\left[u(a) \cdot \left\{ \left(\frac{1}{K(a)} + 1 \right) (1-\alpha) \cdot \frac{(1-\eta)}{\eta} \cdot \frac{1}{1-\Omega(a)} \right\} + (1-\zeta) \right] \check{l}. \end{split}$$

Since

$$\frac{dl}{dg} = \frac{l}{g} \cdot \frac{d\ln(l)}{d\ln(g)} = \frac{1-\zeta}{\zeta} \cdot \frac{\check{l}}{\check{g}},$$

we infer that

$$\frac{dl}{dg} = -\frac{1-\zeta}{1-\zeta+(1-\alpha)\cdot\frac{(1-\eta)}{\eta}\cdot\frac{1+K(a)}{K(a)}\cdot\frac{u(a)}{1-\Omega(a)}}.$$
(A16)

Clearly, $-1 < \partial l/\partial g < 0$ (note that $1 - \Omega(a) > 0$).

A.5 **Proof of Proposition 3**

Using (A16), I infer that

$$\begin{split} \frac{dn}{dg} &= 1 + \frac{dl}{dg} = 1 - \frac{1 - \zeta}{1 - \zeta + (1 - \alpha) \cdot \frac{(1 - \eta)}{\eta} \cdot \frac{1 + K(a)}{K(a)} \cdot \frac{u(a)}{1 - \Omega(a)}} \\ &= \frac{(1 - \alpha) \cdot \frac{(1 - \eta)}{\eta} \cdot \frac{1 + K(a)}{K(a)} \cdot \frac{u(a)}{1 - \Omega(a)}}{1 - \zeta + (1 - \alpha) \cdot \frac{(1 - \eta)}{\eta} \cdot \frac{1 + K(a)}{K(a)} \cdot \frac{u(a)}{1 - \Omega(a)}} \\ &= \frac{1 - \alpha}{1 - \alpha + (1 - \zeta) \cdot \frac{K(a)}{1 + K(a)} \cdot \frac{\eta}{(1 - \eta)} \cdot \frac{1 - \Omega(a)}{u(a)}}. \end{split}$$

I can now express the public-employment multiplier λ_g , which is the increase dn in total employment that can be achieved by spending one unit of private good on public employment. The cost of a marginal increase dg in public employment is $dg \cdot [w + (1 - \delta \cdot (1 - s)) \cdot r \cdot a/q(\theta)] = dg \cdot mpl(a)$. To conclude, the employment multiplier is given by

$$\lambda_g = \frac{1-\alpha}{1-\alpha+(1-\zeta)\cdot\frac{K(a)}{1+K(a)}\cdot\frac{\eta}{(1-\eta)}\cdot\frac{1-\Omega(a)}{u(a)}}\cdot mpl(a)^{-1}.$$

Using Lemma A1, the fact that $q'(\theta) < 0$, as well as the definitions in Section A.1, I infer:

$$\frac{dK(a)}{da} > 0, \ \frac{d[1 - \Omega(a)]}{da} > 0, \ \frac{du(a)}{da} < 0, \ \frac{dmpl(a)}{da} > 0.$$

Hence, $d\lambda_g/da < 0$.

B Solution to the Worker's Problem

The Lagrangian of the worker's problem described in Section 2.3 is

$$\mathbb{E}_0 \sum_{t=0}^{+\infty} \delta^t \left\{ (1-u_t) \cdot \ln(w_t) + u_t \cdot [k(e_t) + \ln(b_t \cdot w_t) + e_t \cdot f(\theta_t) \cdot \ln(1/b_t)] \right. \\ \left. + A_t \cdot \left[u_t \cdot e_t \cdot f(\theta_t) + (1-s) \cdot n_{t-1}^s - n_t^s \right] \right\},$$

where n_t^s is the probability to be employed in period t after history a^t , I define the probability of being unemployed at the beginning of period t: $u_t \equiv 1 - (1 - s) \cdot n_{t-1}^s$, and $\{A_t\}$ is a stochastic process for Lagrange multipliers. The first-order condition with respect to effort e_t gives (after

dividing by $u_t \cdot f(\theta_t)$):

$$\frac{k'(e_t)}{f(\theta_t)} = \ln(1/b_t) + A_t$$

The first-order condition with respect to the probability of being employed n_t^s yields:

$$A_{t} = \delta \cdot (1-s) \cdot \mathbb{E}_{t} \left[k(e_{t+1}) + \left[A_{t+1} + \ln(1/b_{t+1}) \right] \cdot \left[1 - e_{t+1} \cdot f(\theta_{t+1}) \right] \right]$$

Merging the two first-order conditions together yields:

$$\frac{k'(e_t)}{f(\theta_t)} = \ln(1/b_t) + \delta(1-s)\mathbb{E}_t \left[k(e_{t+1}) + \frac{k'(e_{t+1})}{f(\theta_{t+1})} \cdot [1 - e_{t+1}f(\theta_{t+1})] \right]$$

With the isoelastic disutility of effort, $e \cdot k'(e) = (1 + \kappa) \cdot k(e)$, and:

$$\frac{k'(e_t)}{f(\theta_t)} = \ln(1/b_t) + \delta(1-s)\mathbb{E}_t\left[-\kappa k(e_{t+1}) + \frac{k'(e_{t+1})}{f(\theta_{t+1})}\right]$$

Thus, the optimal effort function therefore satisfies the Euler equation (1).

C Equilibrium in a Pissarides [2000] Model

In a Pissarides [2000] model, the stochastic process for wages $\{w_t\}_{t=0}^{+\infty}$ is determined using the generalized Nash solution to the bargaining problem faced by a firm-worker pair. The equilibrium system described in Section 2.5 includes one more equation to characterize the wage process.

Let \mathcal{W}_t denote the equilibrium surplus to a worker from being employed after the matching process in period *t*. Worker's optimal search behavior satisfies $k'(e_t)/f(\theta_t) = \mathcal{W}_t^{20}$ Rewriting the first-order condition (1) determining worker's optimal search effort as

$$\frac{k'(e_t)}{f(\theta_t)} = \ln\left(\frac{1}{b_t}\right) + \delta \cdot (1-s) \cdot \mathbb{E}_t\left[\frac{k'(e_{t+1})}{f(\theta_{t+1})} - \kappa \cdot k(e_{t+1})\right],$$

I infer the the worker's share of the match surplus satisfies

$$\mathcal{W}_{t} = \ln\left(\frac{1}{b_{t}}\right) + \delta \cdot (1-s) \cdot \mathbb{E}_{t}\left[\mathcal{W}_{t+1} - \kappa \cdot k(e_{t+1})\right].$$
(A17)

In equilibrium, the firm's surplus from an established relationship is simply given by the hiring

²⁰Let \mathcal{E}_t and \mathcal{U}_t be the worker's continuation values from being employed and unemployed respectively at time *t*. Then the worker maximizes $-k(e_t) + e_t \cdot f(\theta_t) \cdot \mathcal{E}_t + [1 - e_t \cdot f(\theta_t)] \cdot \mathcal{U}_t$ over effort e_t . The first-order condition with respect to e_t is $k'(e_t) = f(\theta_t) \cdot [\mathcal{E}_t - \mathcal{U}_t] = f(\theta_t) \cdot \mathcal{W}_t$.

cost since a firm can immediately replace a worker at that cost during the matching period:

$$\mathcal{F}_t = \frac{r \cdot a_t}{q(\mathbf{\theta}_t)}.$$

Assume that wages are continually renegotiated. Then the wage bargained in the current period only influences payoffs in the current period. Accordingly, since the firm's utility is simply its profits, a wage w_t brings a utility $-w_t$ to the firm (or its owners) and

$$\frac{d\mathcal{F}_t}{dw_t} = -1.$$

Since workers would consume exactly the wage w_t in the current period (there are no saving and no borrowing), a wage brings utility $\ln(w_t)$ to the employed worker in the current period and

$$\frac{d \mathcal{W}_t}{dw_t} = \frac{1}{w_t}.$$

The generalized Nash solution to the bargaining problem faced by a firm-worker pair is the wage w_t that maximizes

$$\mathcal{W}_t(w_t)^{\boldsymbol{\beta}} \cdot \mathcal{F}_t(w_t)^{1-\boldsymbol{\beta}}$$

 β is the worker's bargaining power. The first-order condition of this maximization problem implies that the worker's surplus each period is related to the firm's surplus

$$\mathcal{W}_t = \frac{\beta}{1-\beta} \cdot \frac{\mathcal{F}_t}{w_t} = \frac{\beta}{1-\beta} \cdot \frac{r \cdot a_t}{q(\theta_t)} \cdot \frac{1}{w_t}.$$
(A18)

Plugging the expression (A18) for the worker's surplus, which reflect surplus sharing, into the recursive equation (A17) for the worker's surplus, which reflects optimal job search, I derive a new relation between equilibrium variables $\{\theta_t, e_t, w_t\}$ and technology $\{a_t\}$ imposed by setting wages using the Nash bargaining solution:

$$\frac{\beta}{1-\beta} \cdot \frac{r}{q(\theta_t)} \cdot \frac{a_t}{w_t} = \ln\left(\frac{1}{b_t}\right) + \delta \cdot (1-s) \cdot \mathbb{E}_t \left[\frac{\beta}{1-\beta} \cdot \frac{r}{q(\theta_{t+1})} \cdot \frac{a_{t+1}}{w_{t+1}} - \kappa \cdot k(e_{t+1})\right].$$

It is useful to express the disutility from job search in equilibrium as a function of other equilibrium

variables:

$$\begin{aligned} k(e_t) &= e_t \cdot \frac{k'(e_t)}{1+\kappa} \\ k(e_t) &= e_t \cdot \frac{1}{1+\kappa} \cdot \frac{k'(e_t)}{f(\theta_t)} \cdot f(\theta_t) \\ k(e_t) &= e_t \cdot \frac{1}{1+\kappa} \cdot \mathcal{W}_t \cdot f(\theta_t) \\ k(e_t) &= \frac{\beta}{1-\beta} \cdot e_t \cdot \frac{1}{1+\kappa} \cdot \frac{r \cdot a_t}{q(\theta_t)} \cdot \frac{1}{w_t} \cdot f(\theta_t) \\ k(e_t) &= \frac{1}{1+\kappa} \cdot \frac{\beta}{1-\beta} \cdot e_t \cdot r \cdot \theta_t \cdot \frac{a_t}{w_t}. \end{aligned}$$

Combining both equations yield

$$\beta \cdot \frac{r}{q(\theta_t)} \cdot \frac{a_t}{w_t} = (1 - \beta) \ln\left(\frac{1}{b_t}\right) + \beta \cdot \delta \cdot (1 - s) \cdot \mathbb{E}_t \left[\frac{a_{t+1}}{w_{t+1}} \cdot \left\{\frac{r}{q(\theta_{t+1})} - \frac{\kappa}{1 + \kappa} \cdot e_{t+1} \cdot r \cdot \theta_{t+1}\right\}\right].$$

In a static environment, this wage equation becomes:

$$\frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right) \cdot \frac{w}{a} = [1-\delta \cdot (1-s)] \frac{r}{q(\theta)} + \delta \cdot (1-s) \cdot \frac{\kappa}{1+\kappa} \cdot e \cdot r \cdot \theta.$$

Combining this equation with the firm's optimal employment decision (10) yields

$$\frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right) \cdot \left[1 - \left[1 - \delta \cdot (1-s)\right] \cdot \frac{r}{q(\theta)}\right] = \left[1 - \delta \cdot (1-s)\right] \cdot \frac{r}{q(\theta)} + \delta \cdot (1-s) \cdot \frac{\kappa}{1+\kappa} \cdot e \cdot r \cdot \theta$$
$$\frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right) = \left[1 - \delta \cdot (1-s)\right] \cdot \frac{r}{q(\theta)} \left[1 + \frac{1-\beta}{\beta} \cdot \ln\left(\frac{1}{b}\right)\right] + \delta \cdot (1-s) \cdot \frac{\kappa}{1+\kappa} \cdot e \cdot r \cdot \theta.$$

D The Effect of Job-Search Monitoring on Employment

D.1 A model with monitoring

Monitoring provides higher incentives to search for a job. In equilibrium, jobseekers exert more effort, which makes makes recruiting cheaper for firms, reduces the marginal cost of labor, leading firms to hire more workers and produce more. Unemployed workers who are searching for a job when monitored receive UI benefits $b_t \cdot w_t$. Unemployed workers who are not searching for a job when monitored receive a punishment $z_t \cdot b_t \cdot w_t$, with $z_t < 1$. In practice, the punishment may be an income-assistance program granted to unemployed workers irrespective of their search effort. Each period an unemployed worker faces a probability π to be monitored. Given that an unemployed worker is not monitored and receives UI benefits $b_t \cdot w_t$; (2) with probability $\pi \cdot e$,

the unemployed worker is monitored while searching and receives UI benefits $b_t \cdot w_t$; and (3) with probability $\pi \cdot (1-e)$, the unemployed worker is monitored while not searching and receives a sanction $z_t \cdot b_t \cdot w_t$. Monitoring *u* unemployed workers for a fraction π of the time has a cost $\pi \cdot u \cdot m$ in units of private goods, where *m* is a parameter.²¹

D.2 Solution to the worker's problem with placement agency

Given the government policy $\{\pi_t, b_t, z_t\}_{t=0}^{+\infty}$, labor market tightness and wage processes $\{\theta_t, w_t\}_{t=0}^{+\infty}$, the *worker's problem* is to choose a stochastic process for search effort $\{e_t\}_{t=0}^{+\infty}$ to maximize the expected utility

$$\mathbb{E}_{0}\sum_{t=0}^{+\infty}\delta^{t}\cdot\{(1-u_{t})\cdot\ln(w_{t})+u_{t}\cdot[-k(e_{t})+\ln(b_{t}\cdot w_{t})+e_{t}\cdot f(\theta_{t})\cdot\ln(1/b_{t})-(1-e_{t})\cdot\pi_{t}\cdot\ln(1/z_{t})]\},$$

subject to the same law of motion of the probability n_t^s to be employed in period t for all t:

$$n_t^s = \left[1 - (1 - s) \cdot n_{t-1}^s\right] \cdot e_t \cdot f(\boldsymbol{\theta}_t) + (1 - s) \cdot n_{t-1}^s.$$

I define the probability of being unemployed at the beginning of period *t*: $u_t \equiv 1 - (1 - s)n_{t-1}^s$. The Lagrangian of the worker's problem is

$$\mathbb{E}_{0} \sum_{t=0}^{+\infty} \delta^{t} \cdot \left\{ (1-u_{t}) \cdot \ln(w_{t}) + u_{t} \cdot [-k(e_{t}) + \ln(b_{t} \cdot w_{t}) + e_{t} \cdot f(\theta_{t}) \cdot \ln(1/b_{t}) - (1-e_{t}) \cdot \pi_{t} \cdot \ln(1/z_{t})] + A_{t} \left[u_{t} \cdot e_{t} \cdot f(\theta_{t}) + (1-s) \cdot n_{t-1}^{s} - n_{t}^{s} \right] \right\}.$$

where n_t^s is the probability to be employed in period t after history a^t , and $\{A_t\}$ is a stochastic process for the Lagrange multiplier. The first-order conditions with respect to effort e_t yields

$$\frac{k'(e_t)}{f(\theta_t)} = \left[\ln\left(\frac{1}{b_t}\right) + \frac{\pi_t}{f(\theta_t)} \cdot \ln\left(\frac{1}{z_t}\right)\right] + A_t.$$
(A19)

The Lagrange multiplier A_t captures the influence of future labor market conditions. The first-order condition with respect to the probability of being employed n_t^s yields

$$A_{t} = \delta(1-s)\mathbb{E}_{t}\left[k(e_{t+1}) + A_{t+1}\left[1 - e_{t+1}f(\theta_{t+1})\right] + \left[1 - e_{t+1}f(\theta_{t+1})\right]\ln\left(\frac{1}{b_{t+1}}\right) + (1 - e_{t+1})\pi_{t+1}\ln\left(\frac{1}{z_{t+1}}\right)\right]$$

 $^{^{21}}$ The resource cost of monitoring includes administrative costs, time spent interviewing jobseekers, or time spent verifying worker's search efforts.

Substituting for A_t and A_{t+1} using (A19), with the isoelastic disutility function $e \cdot k'(e)/(1+\kappa) = k(e)$, I obtain the following optimality condition:

$$\frac{k'(e_t)}{f(\theta_t)} = \ln\left(\frac{1}{b_t}\right) + \frac{\pi_t}{f(\theta_t)}\ln\left(\frac{1}{z_t}\right) + \delta(1-s)\mathbb{E}_t\left[\frac{k'(e_{t+1})}{f(\theta_{t+1})} - \kappa k(e_{t+1}) - \pi_{t+1}\frac{1 - f(\theta_{t+1})}{f(\theta_{t+1})}\ln\left(\frac{1}{z_{t+1}}\right)\right]$$

In a static environment:

$$[1 - \delta \cdot (1 - s)] \cdot \frac{k'(e)}{f(\theta)} + \delta \cdot (1 - s) \cdot \kappa \cdot k(e) = \ln\left(\frac{1}{b}\right) + \pi \cdot \ln\left(\frac{1}{z}\right) \cdot \left[\delta \cdot (1 - s) + \frac{1 - \delta \cdot (1 - s)}{f(\theta)}\right]$$

D.3 Monitoring multiplier

I assume that that there is no job-search monitoring, $\pi = 0$, in the static environment. This assumption captures the existing state of government intervention on the labor market in the US, and simplifies derivations.²²

In a static environment parameterized by technology *a*, given policies $\{g,\pi\}$, the equilibrium $\{\theta, e, l\}$ on the labor market is characterized by three equations: (17), (8), and (10). I report these three equations here for convenience:

$$l+g = \frac{e \cdot f(\theta)}{s + (1-s) \cdot e \cdot f(\theta)}$$
$$\ln(1/z) \left[\delta \cdot (1-s) + \frac{1 - \delta \cdot (1-s)}{f(\theta)} \right] = \left[1 - \delta \cdot (1-s)\right] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1-s) \cdot k(e)$$
$$y'(l) = \frac{w}{a} + \left[1 - \delta \cdot (1-s)\right] \frac{r}{q(\theta)}.$$

I now consider a marginal increase in the monitoring probability π , keeping replacement rates b, z and public employment g constant. I log-linearize this system to obtain the effect of a marginal change $d\pi$ in monitoring probability, around the steady-state value of $\pi = 0$. I start by log-linearizing the worker's optimality condition:

$$d\pi \cdot \ln(1/z) \left[\delta \cdot (1-s) + \frac{1 - \delta \cdot (1-s)}{f(\theta)} \right] = \left[1 - \delta \cdot (1-s) \right] \cdot \frac{k'(e)}{f(\theta)} \cdot \left[\kappa \check{e} - (1-\eta) \cdot \check{\theta} \right] \\ + \kappa \cdot \delta \cdot (1-s) \cdot k(e) \cdot \left[1 + \kappa \right] \check{e}$$

²²The OECD reports expenditure on public employment service, including placement services, for the US for 1985–2008. This expenditure influences the administration capacity to monitor job search effectively. On average only 0.047% of GDP is spent on this service, explaining why many states have switched from in-person filing and regular job-search reporting to reporting by telephone [Grubb, 2011].

Hence, the log-linear system becomes:

$$\begin{aligned} (1-\zeta) \cdot \check{l} &= u(a) \cdot \left[\check{e} + (1-\eta) \cdot \check{\theta}\right] \\ d\pi \cdot \ln(1/z) \left[\delta \cdot (1-s) + \frac{1-\delta \cdot (1-s)}{f(\theta)} \right] &= (1-\delta \cdot (1-s)) \cdot \frac{k'(e)}{f(\theta)} \left[-(1-\eta) \cdot \check{\theta} \right] \\ &+ \kappa \check{e} \cdot \left[(1-\delta \cdot (1-s)) \cdot \frac{k'(e)}{f(\theta)} + \delta \cdot (1-s) \cdot k(e) \cdot [1+\kappa] \right] \\ &(\alpha-1) \cdot \check{l} = \eta \cdot [1-\Omega(a)] \cdot \check{\theta} \end{aligned}$$

where $\zeta = g/n$ is the share of public employment in total employment, $u(a) = 1 - (1 - s) \cdot n(a)$ is unemployment, $\Omega(a)$ is characterized by (A2) and K(a) is characterized by (A3). To derive the monitoring multiplier, I rewrite the log-linear system as:

$$\check{e} = \frac{(1-\zeta)}{u(a)} \cdot \check{l} - (1-\eta) \cdot \check{\theta}$$
$$-(1-\eta)\check{\theta} = (1-\alpha) \cdot \frac{1-\eta}{\eta} \cdot \frac{1}{1-\Omega(a)} \cdot \check{l}$$
$$d\pi \cdot \ln(1/z) \left[\delta \cdot (1-s) + \frac{1-\delta \cdot (1-s)}{f(\theta)} \right] = (1+\kappa) \cdot \ln(1/b) \left[-(1-\eta) \cdot \check{\theta} \right]$$
$$+ \kappa \cdot \check{l} \cdot \frac{1-\zeta}{u(a)} \cdot \left[\ln(1/b) + \delta \cdot (1-s) \cdot k(e) \right]$$

We infer that

$$\frac{dn}{d\pi} = \frac{dl}{d\pi} = l \cdot \frac{\check{l}}{d\pi} = \frac{l \cdot \ln(1/z) \left[\delta \cdot (1-s) + \frac{1-\delta \cdot (1-s)}{f(\theta)}\right]}{(1+\kappa) \cdot \ln(1/b) \left[(1-\alpha) \cdot \frac{1-\eta}{\eta} \cdot \frac{u(a)}{1-\Omega(a)}\right] + \kappa \cdot (1-\zeta) \cdot \left[\ln(1/b) + \delta \cdot (1-s) \cdot k(e)\right]}$$

I can now express the monitoring multiplier λ_{π} , which is the increase dn in total employment that can be achieved by spending one unit of private good on monitoring job search. The cost of a marginal increase $d\pi$ in monitoring is $d\pi \cdot m \cdot u(a)$. Thus the monitoring multiplier is given by

$$\lambda_{\pi} = \frac{\ln(1/z)}{m} \cdot \frac{l \cdot \left[\delta \cdot (1-s) + \frac{1-\delta \cdot (1-s)}{f(\theta)}\right]}{(1+\kappa)(1-\alpha) \cdot \frac{1-\eta}{\eta} \frac{u(a)}{1-\Omega(a)} \ln(1/b) + \kappa(1-\zeta) \left[\ln(1/b) + \delta(1-s) \cdot k(e)\right]}.$$
 (A20)

I can now characterize the cyclicality of the monitoring multiplier over the business cycle.

PROPOSITION A1. Assume $\alpha < 1$ and $\gamma < 1$. Also assume $\delta = 1$. Then there exists $a^* > 0$ such that for all $a < a^*$, $d\lambda_{\pi}/da > 0$.

The assumption on the discount factor δ is a technical assumption to simplify the algebra. The assumption that $\gamma < 1$ is necessary to obtain fluctuations in unemployment and other variables.

With $\gamma = 1$, $d\lambda_{\pi}/da = 0$. The assumption that $\alpha < 1$ is necessary to obtain a downward-sloping aggregate demand for labor.

Proof. I assume that $\delta = 1$. I start from the above expression for the monitoring multiplier λ_{π} . Notice that:

$$\frac{1}{1-\Omega(a)} = \frac{1}{1-\delta\cdot(1-s)} \cdot \frac{q(\theta)}{r} \cdot l^{\alpha-1} = \frac{1}{s\cdot r} \cdot q(\theta) \cdot l^{\alpha-1}.$$

Using the optimal search condition (8) and the isoelasticity of the disutility of effort $k(e) = e \cdot k'(e)/(1+\kappa)$, I find that

$$\ln(1/b) + \delta \cdot (1-s) \cdot k(e) = [1 - \delta \cdot (1-s)] \cdot \frac{k'(e)}{f(\theta)} + (1+\kappa) \cdot \delta \cdot (1-s) \cdot k(e)$$
$$= \frac{1+\kappa}{n} \cdot k(e).$$

Rearranging this equation I infer that

$$k(e) = \ln(1/b) \cdot \frac{n}{\kappa + u}$$
$$\ln(1/b) + \delta \cdot (1 - s) \cdot k(e) = \ln(1/b) \cdot \frac{1 + \kappa}{\kappa + u}.$$

Therefore,

$$\lambda_{\pi} = Q \cdot \frac{(\kappa + u) \cdot l \cdot \left[(1 - s) + \frac{s}{f(\theta)}\right]}{(\kappa + u) \cdot (1 - \alpha) \cdot \frac{1 - \eta}{\eta} \cdot u \cdot \frac{1}{s \cdot r} \cdot q(\theta) \cdot l^{\alpha - 1} + \kappa \cdot (1 - \zeta)}.$$

where

$$Q = \frac{\ln(1/z)}{m} \cdot \frac{1}{(1+\kappa) \cdot \ln(1/b)} > 0$$

is a constant that depends on the parameters of the model. Multiplying numerator and denominator by $e \cdot f(\theta)/l = s/(u \cdot (1 - \zeta))$ yields

$$\lambda_{\pi} = Q \cdot \frac{(\kappa + u) \cdot [(1 - s) \cdot f(\theta) + s] \cdot e}{(\kappa + u) \cdot \frac{1 - \alpha}{1 - \zeta} \cdot \frac{1 - \eta}{\eta} \cdot \frac{q(\theta)}{r} \cdot l^{\alpha - 1} + \kappa \cdot \frac{s}{u}}.$$

First step. I determine how $N = (\kappa + u) \cdot [(1 - s) \cdot f(\theta) + s] \cdot e$ responds to a marginal change *da* in technology. Using the same logic as for the derivation of equations (A14) and (A15), I infer that the log-deviations \check{e} and \check{n} of equilibrium effort and aggregate employment are related to the

log-deviation $\check{\theta}$ of labor market tightness by

$$\check{e} = (1 - \eta) \cdot \frac{1}{K(a)} \cdot \check{\Theta}$$
(A21)

$$\check{n} = u\left[\check{e} + (1-\eta)\cdot\check{\theta}\right] = (1-\eta)\cdot\frac{1+K(a)}{K(a)}\cdot u\cdot\check{\theta}$$
(A22)

$$\check{u} = -\frac{1-u}{u}\check{n} = -(1-u)\cdot(1-\eta)\cdot\frac{1+K(a)}{K(a)}\cdot\check{\theta}$$
(A23)

where the last equality holds because $\check{u} = -(1-u)/u \cdot \check{n}$ by definition of unemployment. I infer that the log-deviation of the numerator of λ_{π} is

$$\begin{split} \check{N} &= \check{e} + (1 - \eta) \cdot \frac{(1 - s) \cdot f(\theta)}{s + (1 - s) \cdot f(\theta)} \cdot \check{\theta} + \frac{u}{u + \kappa} \cdot \check{u} \\ \check{N} &= (1 - \eta) \cdot \check{\theta} \cdot \left\{ \left[\frac{1}{K(a)} + \frac{(1 - s) \cdot f(\theta)}{s + (1 - s) \cdot f(\theta)} \right] - \frac{u \cdot (1 - u)}{u + \kappa} \cdot \left[1 + \frac{1}{K(a)} \right] \right\} \end{split}$$

Note that since $e \leq 1$,

$$\frac{(1-s)\cdot f(\theta)}{s+(1-s)\cdot f(\theta)} = \frac{(1-s)\cdot e\cdot f(\theta)}{e\cdot s+(1-s)\cdot e\cdot f(\theta)}$$
$$\frac{(1-s)\cdot f(\theta)}{s+(1-s)\cdot f(\theta)} > \frac{(1-s)\cdot e\cdot f(\theta)}{s+(1-s)\cdot e\cdot f(\theta)} = (1-s)\cdot n = 1-u > (1-u)\cdot \frac{u}{u+\kappa}$$

Furthermore it is clear that $u \cdot (1-u)/(u+\kappa) < 1$. Hence $\check{N}/\check{\theta} > 0$. But as showed by Lemma A1, $\check{\theta}/\check{a} > 0$. Thus, $\check{N}/\check{a} > 0$ and clearly dN/da > 0.

Second step. I determine how

$$Q = (\kappa + u) \cdot \frac{1 - \alpha}{1 - \zeta} \cdot \frac{1 - \eta}{\eta} \cdot \frac{q(\theta)}{r} \cdot l^{\alpha - 1} + \kappa \cdot \frac{s}{u}$$

responds to a marginal change da in technology. Notice that $\check{l} = \check{n}$. Also notice that since $\delta = 1$, $K(a) = \kappa/u$. Using (A21), (A22), (A23), I find

$$\begin{aligned} \frac{dQ}{\check{\Theta}} &= \kappa \cdot \frac{s}{u} \cdot \left[(1-u) \cdot (1-\eta) \cdot \frac{u+\kappa}{\kappa} \right] \\ &- \left[(\kappa+u) \cdot \frac{1-\alpha}{1-\zeta} \cdot \frac{1-\eta}{\eta} \cdot \frac{q(\theta)}{r} \cdot l^{\alpha-1} \right] \\ &\cdot \left[\eta + (1-\alpha) \cdot (1-\eta) \cdot \frac{u+\kappa}{\kappa} \cdot u + \frac{u}{\kappa+u} \cdot (1-u) \cdot (1-\eta) \cdot \frac{u+\kappa}{\kappa} \right] \\ \frac{\kappa}{(u+\kappa) \cdot (1-u)} \cdot \frac{\theta}{1-\eta} \cdot \frac{dQ}{d\theta} &= \kappa \cdot \frac{s}{u} - \left[\frac{1-\alpha}{1-\zeta} \cdot \frac{1-\eta}{\eta} \cdot \frac{q(\theta)}{r} \cdot l^{\alpha-1} \right] \\ &\cdot \left[\frac{\kappa}{1-u} \cdot \frac{\eta}{1-\eta} + (1-\alpha) \cdot (u+\kappa) \cdot \frac{u}{1-u} + u \right] \end{aligned}$$

Unfortunately, I cannot sign $dQ/d\theta$ for any u (or any underlying a). However Lemma A1 says that 1/u is increasing with a while $q(\theta), l^{\alpha-1}, 1/(1-u), u, u(u+\kappa)/(1-u)$ are all decreasing with a. Hence the right-hand side of the equation above is increasing with a. When $a \to 0, n \to 0$, $u \to 1, \theta \to 0$, so the right-hand side $\to -\infty$. Accordingly, there exists $a^* > 0$ such that for all $a < a^*$, the right-hand side is negative, and $dQ/d\theta < 0$. Since, from Lemma A1, $d\theta/da > 0$, I infer that there exists $a^* > 0$ such that for all $a < a^*, dQ/da < 0$. In that case d(N/Q)/da > 0 and $d\lambda_{\pi}/da > 0$.

D.4 Calibration

Transfers from the government. If unemployed workers do not actively search for a job, they lose their entitlement to unemployment benefits. I assume that in that case, the only social assistance provided is food stamps.²³ Using a detailed state-level data, Pavoni and Violante [2007] compute that in 1996 on a monthly basis, the median monthly allotment of food stamps for a family of four was \$397. At the same time, the median monthly wage paid to a worker with at most a high-school diploma (who would be eligible for food stamps) is \$1,540. Hence, I set $z \times b = 397/1,540 = 26\%$. Hence, z = 0.26/0.60 = 43%.

Monitoring cost function. The US Department of Education and the US Department of Health and Human Services jointly sponsored a large-scale evaluation of welfare-to-work policies, the National Evaluation of Welfare-to-Work Strategies, based on data pertaining to over 40,000 individuals followed for 5 years in the 1991–1999 period, in 11 separate locations [Hamilton et al.,

²³As explained by Pavoni and Violante [2007], once unemployment insurance benefits and other job search assistance programs expire, households remain virtually without any other form of benefits and have the right to the maximum allotment of food stamps. The Food Stamp program provides monthly coupons to eligible low-income families, which can be used to purchase food.

2001]. In particular, in three locations, Atlanta, GA, (1,441 treated individuals), Grand Rapids, MI (1,557 individuals treated individuals), and Riverside, CA (3,384 individuals treated individuals), implemented labor force attachment programs in which monitored job search was the first mandatory activity. In the three sites considered, these welfare-to-work program requires unemployed workers to participate to a minimum of 20 hours of monitored job search per week. Given a normal work week of 40 hours, this corresponds to a monitoring probability of $\pi = 0.5$, as they are monitored half of the time. The average cost per unemployed workers per month of participation across the three sites is $1/3 \cdot [416 + (259 + 1025) + 759] = 820$ in 1999 \$.²⁴ The average weekly earning for production workers from the CES for 1999 in 1999 \$ is 464\$, which gives an average monthly earning of 1856\$. Hence the monitoring cost for a monitoring probability of $\pi = 0.5$ is 820/1856 = 44% of average earning. Hence the marginal monitoring cost *m* satisfies $m \cdot 0.5 = 0.44 \cdot \omega$, which implies $m = 0.88 \cdot \omega = 0.63$.

E The Effect of Hiring Subsidies on Employment

E.1 Wage-subsidy multiplier

In a static environment parameterized by technology *a*, given policies $\{g, b, \tau\}$, the equilibrium $\{\theta, e, l\}$ on the labor market is characterized by three equations: (17), (8), and (10) (modified to take the wage subsidy into account). I report these three equations here for convenience:

$$l+g = \frac{e \cdot f(\theta)}{s + (1-s) \cdot e \cdot f(\theta)}$$
$$\ln(1/b) = [1 - \delta \cdot (1-s)] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1-s) \cdot k(e)$$
$$y'(l) = (1-\tau) \cdot \frac{w}{a} + [1 - \delta \cdot (1-s)] \cdot \frac{r}{q(\theta)}.$$

I now consider a marginal increase in the wage subsidy, keeping public employment g and replacement rate b constant. I log-linearize this system to obtain the effect of a marginal change in wage subsidy $d\tau$, around an equilibrium with $\tau = 0$ (so that the equilibrium is comparable to the equilibria around which I performed the previous policy experiments):

$$(1 - \zeta) \cdot \mathring{l} = u(a) \cdot [\check{e} + (1 - \eta) \cdot \check{\theta}]$$

$$K(a) \cdot \check{e} = (1 - \eta) \cdot \check{\theta}$$

$$(\alpha - 1) \cdot \check{l} = \eta \cdot [1 - \Omega(a)] \cdot \check{\theta} - \Omega(a) \cdot d\tau$$

²⁴These costs include: operating costs (overheads, space rental, case managers expenditure, job search facilitation, classroom instruction), costs of support service to enable participation (for instance, childcare for single parents).

where $\zeta = g/n$ is the share of public employment in total employment, $\Omega(a)$ is characterized by (A2) and K(a) is characterized by (A3). I rewrite the log-linear system as:

$$\begin{split} \check{e} &= (1-\eta) \cdot \frac{1}{K(a)} \cdot \check{\Theta} \\ \check{\Theta} &= \frac{1-\zeta}{u(a) \cdot (1-\eta) \cdot (1+\frac{1}{K(a)})} \cdot \check{l} \\ \Omega(a) \cdot d\tau &= \left[(1-\alpha) + \eta \cdot [1-\Omega(a)] \frac{1-\zeta}{u(a) \cdot (1-\eta)(1+\frac{1}{K(a)})} \right] \cdot \check{l}. \end{split}$$

Since $dn/d\tau = dl/d\tau = l \cdot \check{l}/d\tau$, I infer that

$$\frac{dn}{d\tau} = l \cdot \frac{\Omega(a)}{(1-\alpha) + (1-\zeta) \cdot \frac{1-(1-\tau) \cdot \Omega(a)}{u(a)} \cdot \frac{\eta}{1-\eta} \cdot \frac{K(a)}{1+K(a)}}$$

I can now express the wage-subsidy multiplier λ_{τ} , which is the increase *dn* in total employment that can be achieved by spending one dollar on a wage subsidy. The cost of a marginal increase in wage subsidy is $d\tau \cdot l \cdot w(a)$. Hence, the multiplier is given by

$$\lambda_{\tau} = \frac{1}{(1-\alpha) + (1-\zeta) \cdot \frac{1-\Omega(a)}{u(a)} \cdot \frac{\eta}{1-\eta} \cdot \frac{K(a)}{K(a)+1}} \cdot mpl(a)^{-1} = \frac{1}{1-\alpha} \cdot \lambda_g.$$

E.2 Equivalence of wage subsidy with recruiting subsidy

In this section, I briefly relate a wage subsidy to a recruiting subsidy, also commonly used in OECD countries. I assume that the government covers a fraction $\vartheta \in [0, 1]$ of the recruiting cost $r \cdot a/q(\theta)$ paid by the firm for each new hire. In a static environment parameterized by technology a, given policies $\{g, b, \vartheta\}$, the equilibrium $\{\theta, e, l\}$ on the labor market is characterized by three equations: (17), (8), and (10) (modified to take the recruiting subsidy into account). I report these three equations here for convenience:

$$l+g = \frac{e \cdot f(\theta)}{s + (1-s) \cdot e \cdot f(\theta)}$$
$$\ln(1/b) = [1 - \delta \cdot (1-s)] \cdot \frac{k'(e)}{f(\theta)} + \kappa \cdot \delta \cdot (1-s) \cdot k(e)$$
$$y'(l) = \frac{w}{a} + (1-\vartheta) \cdot [1 - \delta \cdot (1-s)] \frac{r}{q(\theta)}.$$

I now consider a marginal increase in the recruiting subsidy, keeping public employment g and replacement rate b constant. I log-linearize this system to obtain the effect of a marginal change in wage subsidy $d\vartheta$, around an equilibrium with $\vartheta = 0$ (so that the equilibrium is comparable to the

equilibria around which I performed the previous policy experiments):

$$(1-\zeta) \cdot \check{l} = u(a) \cdot \left[\check{e} + (1-\eta) \cdot \check{\theta}\right]$$
$$K(a) \cdot \check{e} = (1-\eta) \cdot \check{\theta}$$
$$(\alpha-1) \cdot \check{l} = [1-\Omega(a)] \cdot \left[\eta \cdot \check{\theta} - d\vartheta\right]$$

where $\zeta = g/n$ is the share of public employment in total employment, $\Omega(a)$ is characterized by (A2) and K(a) is characterized by (A3). It is convenient to rewrite the log-linear system as:

$$\check{e} = (1-\eta) \cdot \frac{1}{K(a)} \cdot \check{\theta}$$
$$\check{\theta} = \frac{1-\zeta}{u(a) \cdot (1-\eta) \cdot (1+\frac{1}{K(a)})} \cdot \check{l}$$
$$[1-\Omega(a)] \cdot d\vartheta = \left[(1-\alpha) + \eta \cdot [1-\Omega(a)] \frac{1-\zeta}{u(a) \cdot (1-\eta)(1+\frac{1}{K(a)})} \right] \cdot \check{l}.$$

Since $dn/d\vartheta = dl/d\vartheta = l \cdot \check{l}/d\vartheta$, I infer that

$$\frac{dn}{d\vartheta} = l \cdot \frac{[1 - \Omega(a)]}{(1 - \alpha) + (1 - \zeta) \cdot \frac{1 - \Omega(a)}{u(a)} \cdot \frac{\eta}{1 - \eta} \cdot \frac{K(a)}{1 + K(a)}}.$$

I can now express the recruiting-subsidy multiplier λ_{ϑ} , which is the increase dn in total employment that can be achieved by spending one dollar on a recruiting subsidy. The cost of a marginal increase in wage subsidy is $d\vartheta \cdot [r \cdot a/q(\theta)] \cdot [s \cdot l]$, which can rewritten $d\vartheta \cdot [mpl(a) \cdot \{1 - \Omega(a)\}] \cdot [s/(1 - \delta \cdot (1 - s))] \cdot l$. Hence, the multiplier is given by

$$\lambda_{\vartheta} = \frac{\left[1 - \delta \cdot (1 - s)\right]/s}{(1 - \alpha) + (1 - \zeta) \cdot \frac{1 - \Omega(a)}{u(a)} \cdot \frac{\eta}{1 - \eta} \cdot \frac{K(a)}{1 + K(a)}} \cdot mpl(a)^{-1}.$$

Hence, wage-subsidy and recruiting-subsidy multipliers are related by a very simple relationship:

$$\lambda_{\vartheta} = rac{1 - \delta \cdot (1 - s)}{s} \cdot \lambda_{\sigma}.$$

If $\delta \to 1$, $1 - \delta \cdot (1 - s) \to s$ and $\lambda_{\vartheta} \to \lambda_{\sigma}$.